

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Hönow, Nils Christian; Karki, Kiran; Burger, Maximilian N.

Working Paper Natural disaster experience does not affect environmental attitudes or prosociality: Evidence from the 2021 flood in Germany

Ruhr Economic Papers, No. 1147

Provided in Cooperation with: RWI – Leibniz-Institut für Wirtschaftsforschung, Essen

Suggested Citation: Hönow, Nils Christian; Karki, Kiran; Burger, Maximilian N. (2025) : Natural disaster experience does not affect environmental attitudes or prosociality: Evidence from the 2021 flood in Germany, Ruhr Economic Papers, No. 1147, ISBN 978-3-96973-330-1, RWI - Leibniz-Institut für Wirtschaftsforschung, Essen, https://doi.org/10.4419/96973330

This Version is available at: https://hdl.handle.net/10419/314410

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



RUHR ECONOMIC PAPERS

Nils Christian Hoenow Kiran Karki Maximilian N. Burger

> Natural Disaster Experience Does Not Affect Environmental Attitudes or Prosociality – Evidence from the 2021 Flood in Germany

> > **CVI** #1147

Imprint

Ruhr Economic Papers

Published by

RWI – Leibniz-Institut für Wirtschaftsforschung Hohenzollernstr. 1-3, 45128 Essen, Germany Ruhr-Universität Bochum (RUB), Department of Economics Universitätsstr. 150, 44801 Bochum, Germany Technische Universität Dortmund, Department of Economic and Social Sciences Vogelpothsweg 87, 44227 Dortmund, Germany Universität Duisburg-Essen, Department of Economics Universitätsstr. 12, 45117 Essen, Germany

Editors

Prof. Dr. Thomas K. Bauer RUB, Department of Economics, Empirical Economics Phone: +49 (0) 234/3 22 83 41, e-mail: thomas.bauer@rub.de Prof. Dr. Ludger Linnemann Technische Universität Dortmund, Department of Business and Economics Economics – Applied Economics Phone: +49 (0) 231/755-3102, e-mail: Ludger.Linnemann@tu-dortmund.de Prof. Dr. Volker Clausen University of Duisburg-Essen, Department of Economics International Economics Phone: +49 (0) 201/1 83-3655, e-mail: vclausen@vwl.uni-due.de Prof. Dr. Ronald Bachmann, Prof. Dr. Almut Balleer, Prof. Dr. Manuel Frondel, Prof. Dr. Ansgar Wübker RWI, Phone: +49 (0) 201/81 49-213, e-mail: presse@rwi-essen.de

Editorial Office

Sabine Weiler

RWI, Phone: +49 (0) 201/81 49-213, e-mail: sabine.weiler@rwi-essen.de

Ruhr Economic Papers #1147

Responsible Editor: Manuel Frondel

All rights reserved. Essen, Germany, 2025

ISSN 1864-4872 (online) - ISBN 978-3-96973-330-1

The working papers published in the series constitute work in progress circulated to stimulate discussion and critical comments. Views expressed represent exclusively the authors' own opinions and do not necessarily reflect those of the editors.

Ruhr Economic Papers #1147

Nils Christian Hoenow, Kiran Karki, and Maximilian N. Burger

Natural Disaster Experience Does Not Affect Environmental Attitudes or Prosociality – Evidence from the 2021 Flood in Germany



Bibliografische Informationen der Deutschen Nationalbibliothek

The Deutsche Nationalbibliothek lists this publication in the Deutsche Nationalbibliografie; detailed bibliographic data are available on the Internet at http://dnb.dnb.de

RWI is funded by the Federal Government and the federal state of North Rhine-Westphalia.

https://dx.doi.org/10.4419/96973330 ISSN 1864-4872 (online) ISBN 978-3-96973-330-1 Nils Christian Hoenow, Kiran Karki, and Maximilian N. Burger*

Natural Disaster Experience Does Not Affect Environmental Attitudes or Prosociality – Evidence from the 2021 Flood in Germany

Abstract

In July 2021, severe floods devastated parts of Germany, causing numerous casualties and extensive damage to property and infrastructure. As climate change is expected to increase the frequency and severity of such extreme weather events, understanding their social implications is crucial. Using data from three nationwide surveys, we examine the impact of the 2021 flood on environmental attitudes, pro-environmental behaviors, and the support for climate-related policies across a wide range of indicators. Results reveal no statistically significant effects, regardless of the estimation methods or measures of flood exposure used. We additionally investigate the flood's effect on prosociality, assessed through measures such as past charitable donations and incentivized decisions in a dictator game. Similarly, we find only limited variation in prosociality, but with impacts differing based on whether respondents in affected areas also sustained damage to their households. These findings challenge the expectation that direct exposure to natural disasters increases environmental awareness and prosocial behavior.

JEL-Codes: D64, D91, Q54, Q58

Keywords: Natural disaster; flood; environmental awareness; environmental attitudes; environmental behavior; climate policy; prosociality

March 2025

^{*} Nils C. Hoenow, RWI; Kiran Karki, RWI; Maximilian N. Burger, Department of Economics, Philipps University Marburg. - We are grateful for helpful comments and ongoing support by the team members at the Department for Environment and Resources at the RWI - Leibniz Institute for Economic Research, in particular Mark Andor, Jana Eßer, Manuel Frondel, Sven Hansteen, Viola Helmers, Eva Hümmecke, Leonie Matejko, Delia Niehues and Lukas Tomberg. We further thank the participants of the GEM/RWI Workshop on Behavioral Environmental Economics (BEE) for their feedback and input at the Grenoble École de Management in February 2024. Funding: We gratefully acknowledge financial support by the Stiftung Mercator, without which this project would not have been possible. Among others, its funding enabled the creation of the RWI Climate-Mobility Panel, the primary data source for this research. Moreover, this research partially builds on data from projects funded by the E.ON Foundation (Socio-Ecological Panel) and the European Commission under the Horizon 2020 Research and Innovation Program, Grant agreement number 837752 (Newcomers). – All correspondence to: Nils C. Hoenow, RWI, Hohenzollernstraße 1–3, 45128 Essen, Germany, e-mail: christian.hoenow@rwi-essen.de

1 Introduction

For years, scholars have warned that human-induced climate change will increase the frequency and intensity of extreme weather events, such as floods, with significant consequences for both, developing and industrialized countries (e.g., Cornwall, 2021; Hirabayashi et al., 2013; IPCC, 2021; Netzel et al., 2021; Reichstein et al., 2021). In July 2021, a catastrophic flood hit Germany, Belgium, Luxembourg and neighboring countries, severely affecting the German states of North Rhine-Westphalia and Rhineland-Palatinate. Despite floods being the most common type of disaster worldwide (CRED, 2020) with Europe being no exception (European Environment Agency, 2024), the scale and impact of the 2021 flood were unparalleled in recent decades (Fekete & Sandholz, 2021). It resulted in the highest flood-related death toll in Germany for the past 60 years (Dietze et al., 2022) with at least 200 fatalities and substantial damage to property and infrastructure (Tradowsky et al., 2023) estimated to amount to 30 billion euro (Apel et al., 2022). Given the increasing frequency of extreme weather events, it is important to understand their broader effects on environmental attitudes, behaviors, and prosociality.

We hypothesize that such extreme events, threatening life, property and livelihood may alter environmental attitudes and pro-environmental behaviors of those affected. While frequent media coverage links extreme events to anthropogenic climate change, many perceive these events as distant and unrelated to their own lives (Spence et al., 2011; Swim et al., 2009). Experiencing a disaster firsthand may reduce this perceived distance, leading to an increased risk perception and awareness of climate change's impacts (Demski et al., 2017; Lorenzoni & Pidgeon, 2006; Marx et al., 2007; McDonald et al., 2015; Rüttenauer, 2024; Weber, 2010). Such shifts in perception could lead to the intention to adopt more environmentally friendly behaviors with the aim of mitigating climate change and limiting future impacts, for example through the reduction of personal greenhouse gas emissions (Haden et al., 2012; Spence et al., 2011; Van Der Linden, 2014; Whitmarsh, 2008). Besides potential changes in individual mitigation efforts, the flood experience may also affect citizens' support for climate-friendly policy measures and lead to according political party preferences or even political involvement (Baccini & Leemann, 2021; Garside & Zhai, 2022; Hilbig & Riaz, 2024; Holub & Schündeln, 2023; Rüttenauer, 2024). Understanding these effects is important for shaping future environmental policy and encouraging behavioral change at both, the individual and societal levels.

In addition to environmental attitudes and behaviors, we hypothesize that experiencing a natural disaster may affect prosociality. The German floods of 2021 led to an extraordinary public response, with donations reaching approximately 655 million euros – the third highest donation amount related to a natural disaster in Germany (Deutsches Zentralinstitut für soziale Fragen, 2022).

Prosocial behavior, however, extends beyond monetary contributions and includes actions such as helping, sharing, and volunteering among fellow citizens (Brief & Motowidlo, 1986). Given that institutional help can often be slow and insufficient, citizens' willingness to engage in prosocial actions is crucial for resilience during crises (Steimanis & Vollan, 2022). In the context of natural disasters, affected individuals and households often receive help from others during the event (e.g., rescue from the floods), in the immediate aftermath (e.g., supply with food and water or removal of debris), as well as in the phase of reconstruction (e.g., donations in kind and money). Understanding how natural disasters influence prosociality is vital for fostering stronger social networks and enhancing collective resilience, particularly in the face of future climate-related crises.

There are several pathways through which we expect that the experience of natural disasters could affect prosociality: First, interactions with others during the event or the recovery phase may create strong social bonds among affected individuals, which can promote prosociality (Cassar et al., 2017). Second, having experienced an extreme event, such as a natural disaster, may enhance one's ability to take the perspective of others in similar situations. Being able to put oneself in the situation of others may increase empathy and the understanding of being in need, thereby enhancing prosociality (Lehmann et al., 2022; Lim & DeSteno, 2016). Even if not personally affected by any damage, being close to others and witnessing their physical and emotional distress can evoke empathy, which may in turn increase prosocial behavior (Decety et al., 2016; Depow et al., 2021; Eisenberg & Miller, 1987; Schroeder et al., 2015). Further, receiving help of any kind during or after a disaster may shift one's view of others, reinforcing the belief that people are generally trustworthy and benevolent, encouraging reciprocal prosocial behavior, possibly lastingly beyond the event itself (Cassar et al., 2017). Finally, the experience of being dependent on the help of others during and after the disaster, coupled with the expectation of experiencing further events in the future, can lead to the strengthening of prosociality to build social networks as safety nets in times of need. Opposingly though, there are also mechanisms that could contribute to a negative association between experiencing natural disasters and prosociality: After a disaster, individuals, whose property or livelihoods have been damaged or whose physical or mental health has been affected, may face constraints in time, dedication and financial resources to engage in prosocial actions and are busy coping with their own damage. Moreover, negative experiences during a crisis - such as a lack of external assistance, conflicts over scarce resources, embezzlement of aid or even looting - can lead to disappointment and negative reciprocity, reducing prosociality (Fleming et al., 2014). While the latter often receives more attention in the media and public perception, the current literature highlights the potential of environmental hazards and natural disasters to strengthen prosociality (Dussaillant & Guzmán, 2014; Rodríguez et al., 2006; Solnit, 2009). However, despite

a surge in analyses published on the matter in the last ten years, the relation between experiencing natural disasters and prosociality remains elusive, with studies finding a negative, positive, or no relation at all. The variance in outcomes suggests that the relationship is contingent upon the specific context. The conditions preceding the event, coupled with the circumstances experienced by those affected during and after the disaster, may affect this relation (Fleming et al., 2014). For a better understanding of the mechanism behind it, it is therefore necessary to examine multiple and diverse conditions, contexts, types of disasters and, not least, populations.

Our study contributes to the literature by investigating the effects of the 2021 flood experience on both environmental attitudes and behaviors, and prosociality in a high income, industrialized country context with strong institutions and high pre-existing levels of awareness for manmade climate change and its link to extreme weather events (Lee et al., 2015; Mewes et al., 2024). Using data from three large-scale nationwide surveys, each conducted within a year of the disaster, we analyze responses from 13,754 participants from both exposed and nonexposed regions. Our findings suggest that the flood experience did not significantly affect environmental attitudes and behaviors. While we furthermore observe no overall effect on prosociality, we identified differences based on whether personal damage was caused by the flood. Specifically, those reporting personal damage donated on average less money to charity, while those living in affected areas without sustaining any damages self-report higher amounts. Further, affected individuals who received any form of aid during or after the flood show higher scores for positive reciprocity compared to those who suffered flood damage but did not receive any aid. These findings suggest that while the flood experience did not substantially alter environmental attitudes and behaviors, it may have had a nuanced impact on prosociality.

The remainder of the paper is organized as follows: Section 2 reviews the literature on the effect of environmental disasters on environmental attitudes and behaviors, as well as on prosociality. The data and methods employed in the analysis are described in section 3. Section 4 presents our results and section 5 discusses them. Finally, section 6 concludes.

2 Discussion of the Literature

2.1 Environmental Attitudes and Behaviors

A growing body of research explores the relationship between natural disaster experience and environmental attitudes and behaviors. Despite the already large number of studies and the diverse contexts investigated, there is no consensus on how natural disaster experiences influence environmental attitudes (for a meta-analysis see Hornsey et al., 2016 and for recent literature reviews see Howe et al., 2019 and Sisco, 2021). Howe et al. (2019) attribute this lack of agreement to methodological variations across studies, such as differing measures for climate and weather trends, as well as survey items and sample selection. Similarly, Sisco (2021) underscores these challenges and expands upon potential mechanisms and contextual factors that may shape the relationship between disaster experience and environmental attitudes. Here, we focus on the most relevant and more recent studies examining the effect of floods on environmental attitudes and behaviors, particularly those in industrialized countries.

In a recent cross-country analysis, Ogunbode et al. (2022) found no link between disaster experience and climate anxiety in a study of 32 countries, suggesting that political orientation and coping capacity may moderate outcomes. Similarly, Spektor et al. (2023) found that while climate event experiences in seven Latin American countries increased the belief in the existence of climate change, they had no effect on anthropogenic attribution or perceived consequences. Peisker (2023) combined survey data from the European Social Survey and national weather data of 206 European regions and reports negligible effects of extreme weather experiences on climate beliefs.

Studies focusing on single countries offer additional nuanced insights. Demski et al. (2017) found greater salience of climate change, elevated intentions for mitigation and more support for climate policies in exposed citizens after a storm with floodings in the United Kingdom. For England and Wales, Lohmann and Kontoleon (2023) found increased risk perception but no impact on environmental attitudes and behaviors due to extreme weather events using a difference-in-differences approach. Rüttenauer (2024) combined panel and flood data from the United Kingdom and found that only close proximity to a flood (2 km or less) significantly affected climate beliefs, with stronger effects among right-leaning individuals and climate skeptics. In Germany, studies on the 2013 flood revealed increased climate risk perceptions, particularly among those living in close proximity (within 1 km radius) to the event (Frondel et al., 2017; Osberghaus & Fugger, 2022). Notably, these effects were most pronounced among individuals who suffered personal damage.

With the link between attitudes and behaviors being well-established (see for example Klöckner, 2013), research increasingly examines whether disaster experiences translate into pro-

environmental actions. Hoffmann et al. (2022) combined weather data for temperature anomalies, heat waves, and dry spells with data from Eurobarometer on environmental concern (34 countries) and Green Parties voter turnout in the European Parliament (28 countries). They found that extreme weather events across Europe were associated with increased environmental concern and higher Green Party votes. However, behavioral responses often seem to depend on event attribution. Ogunbode et al. (2019) showed that flood experiences in the United Kingdom only increased mitigation intentions among those attributing the events to climate change. Attribution in turn was higher among participants with higher pre-existing climate change beliefs and leaning towards the political left. Baccini and Leemann (2021) reported similar findings for Switzerland, making use of the country's unique system of frequent citizen votes on policies. Their study found that recent flood exposure increased support for climate policies. However, this effect diminished significantly within ten months. Similarly, Osberghaus and Demski (2019) explored behavioral responses in Germany following the 2013 flood by combining data on internet searches for green energy options with insurance payouts triggered by the disaster. The authors found that while Germany's 2013 flood spurred searches for green energy, this effect was absent in severely affected regions, possibly due to resource constraints or psychological denial.

The exceptionally severe 2021 flood in Europe has also already been subject to research. Bulut and Samuel (2024) investigated the impact from the perspective of Luxembourg. During the flood, the authors were in the process of collecting data from young people aged 16 - 29, allowing them to compare environmental attitudes before and after the event. Their findings revealed that participants living close to flooded areas exhibited a significant increase in pro-environmental attitudes; however, this shift translated into only limited changes in pro-environmental behavior. In the Netherlands, Duijndam et al. (2023) show that individuals whose homes were flooded were more inclined to adopt in-situ adaptation measures and demonstrated higher intentions to migrate. In contrast, individuals living in affected towns who were not personally harmed, displayed no significant increase in such intentions, highlighting the critical role of direct, personal experience in driving behavioral change.

Studies on German election outcomes suggest mixed effects of flood exposure on Green Party voting during the federal elections held ten weeks after the disaster (Garside and Zhai, 2022; Hilbig and Riaz, 2024; Holub and Schündeln, 2023). Using a binary classification of regions as exposed or non-exposed, Garside and Zhai (2022) as well as Hilbig and Riaz (2024) found that Green Party voting was only slightly elevated in strongly affected regions compared to unaffected ones. Conversely, Holub and Schündeln (2023) used the share of buildings damaged in each municipality as an explanatory variable. Their analysis revealed a significant positive effect of flood damage on

voter turnout for the Green Party, suggesting that the degree of local impact plays a crucial role in influencing electoral outcomes.

Overall, these findings emphasize the variability of outcomes as well as the critical role of the specific contexts, individual factors and their complex interplay, suggesting that further research is needed to disentangle relevant mechanisms and moderating factors (Howe et al., 2019; Sisco, 2021). With our study, we contribute through analyzing three large-scale nationwide data sets that were collected as individual-level surveys. In one data set, we explicitly included questions on the affectedness and damage caused by the 2021 flood. Further, we elicit environmental attitudes as well as behaviors and the support for climate policies across a large range of indicators, not only through self-reported measures but also by an incentivized pro-environmental donation experiment.

2.2 *Prosociality*

Research on the impact of natural disasters on prosociality has surged over the past decade. Despite improved data availability, significant uncertainties about this relationship persist, suggesting that effects are context specific. While many studies focus on low-income countries, evidence from high-income settings remains sparse.

As detailed in section 1, effects of natural disasters on prosociality can, in theory, be positive or negative. We first present a number of studies that have identified *increases* in prosociality. For instance, Méon and Verwimp (2022) observed elevated donation rates in Belgian municipalities affected by a heavy storm. Similarly, Shupp et al. (2017) found heightened trust in tornado-affected cities, particularly towards police, fire departments, and friends, compared to a control city. In Japan, exposure to the Kobe Earthquake in 1995 was linked to increased community volunteering (Yamamura, 2016). Studies in low-income countries have also reported increased prosociality following disasters, such as elevated trust and cooperation after earthquakes in Indonesia (Bai & Li, 2021) and Chile (Calo-Blanco et al., 2017), or floods in Pakistan (Ahmad & Younas, 2021) and a tsunami in Thailand (Cassar et al., 2017).

A second strand of literature argues that natural disasters lead to *decreases* in prosociality. For example, a study using European Social Survey data and EM-DAT data for 12 cases within European countries found that extreme heat waves correlated with decreased trust, although similar effects were not observed in other high-income contexts. In low-income countries, several studies have documented negative impacts of environmental hazards on prosociality. Field experiments conducted after the 2004 tsunami in Sri Lanka (Becchetti et al., 2017), the 2014 Cambodian flood (Fiala, 2017), and the 2010 earthquake in Chile (Fleming et al., 2014) found reduced contributions

in dictator games and trust games post-disaster. Survey studies in the Philippines and Bangladesh further support these findings, showing decreased reciprocity and trust after the 2013 typhoon (Biener & Landmann, 2023) and 1998 flood (Rahman et al., 2020).

A third body of literature finds no significant effect of natural disasters on procosiality. However, to this point and to the best of our best knowledge, zero effects have only been documented in lowincome countries. Studies employing incentivized tasks conducted after events like the 2012 Philippine typhoon (Abatayo & Lynham, 2020), the 2009 Bangladeshi cyclone (Ahsan, 2014), the 2012 Philippine flood (Kuroishi & Sawada, 2019), and the 2011 Cambodia mega flood (Chantarat et al., 2019) showed no effect on prosocial behavior. Similarly, survey studies in Chile and Pakistan found no significant change in trust or prosocial behaviors like donations or volunteering after major earthquakes (Andrabi & Das, 2017; Calo-Blanco et al., 2017; Maki et al., 2019).

Similar to the association with environmental attitudes and behaviors, the variance in outcomes suggests that characteristics of the disaster affecting in-event as well as post-event experience and other contextual factors may be decisive for its impact on prosociality. Few studies explored the mechanisms leading to an alteration of prosociality as a result of such experience. Vardy and Atkinson (2019) observed that suffering personal damage decreases prosocial behavior, while only witnessing others in distress increases it. Another factor that seems to be consistently associated with a positive effect on prosociality is whether affected individuals have received any form of aid (Andrabi & Das, 2017; Becchetti et al., 2017; Cassar et al., 2017; Rahman et al., 2020). Our study provides a new study background in an industrialized country setting and tests through a range of markers of prosociality. We further make use of detailed distinctions in flood experiences, such as whether damage was sustained by the household or any form of aid received, to disentangle the complex pathways through which disasters may influence prosociality.

3 Data and Methods

3.1 Data

For this study, we employ data from three distinct nationwide online surveys conducted within a year after the July 2021 flood event. All three data collections aimed at retrieving samples representative of the German adult population. Table A.1 in the Supplementary Information shows average socioeconomic characteristics of participants from each data set and compares these against the general German population, using official, publicly available data from the German *Mikrozensus*.

The primary survey (data set 1) was conducted eight months after the flood, that is, between March and April 2022, via the survey institute *forsa*. In this survey, 6,285 participants were interviewed towards their perception of climate change, their (self-reported) behavior regarding the environment, their acceptance of climate-related policy measures and their political party preference. It was conducted as wave 3 of the *RWI Climate-Mobility Panel* which is maintained by the RWI – Leibniz Institute for Economic Research within the *Creating the Mobility Transition in Germany together!* research project funded by Stiftung Mercator. While a major part of the survey revolves around mobility behavior and environment-related policy preferences, it also elicits prosociality via hypothetical and incentivized measures. It is the only data set that contains information on prosociality used in this study. Further, only this survey contains direct questions about the flood experience, such as whether participants suffered damage or received aid. The study was preregistered¹ with the intention of focusing the analysis on this first data set. Data sets 2 and 3 were added later to complement with additional information on environmental attitudes and behaviors.

Data set 2 originates from a survey conducted via forsa in July and August 2022, roughly one year after the flood. This survey involved with 6,003 participants and was collected within the *Socio-Ecological Panel* (shortened as *Green SOEP*), also maintained by the RWI and funded by E.ON foundation. It primarily focused on citiziens' perceptions of environmental and energy-related policies, including attitudes and behaviors regarding the environment as well as the support for climate policies.

Data set 3 was drawn from a Europe-wide survey that was conducted within the *NEWCOMERS* research project, funded under the European Union's *Horizon 2020* research and innovation program. For this study, only the German subsample of 1,466 participants is considered. We conducted this survey via *Respondi* and partners (now *Bilendi & Respondi*) four months after the

¹ <u>https://aspredicted.org/jd9pz.pdf</u>

flood in November and December 2021. Its original focus was on citizens' perceptions of energyrelated issues. Relevant to this study, it also included questions on environmental attitudes and behaviors, support for selected climate policies, and an incentivized donation experiment with reallife environmental implications. For this data set, participants were selected to fulfill countryrepresentative quotas, regarding age, gender, income and education. Only adults up to the age of 69 were included.

Table 1 illustrates the variables of interest and their availability across data sets. The exact survey questions for relevant variables are attached in the Supplementary Information (Q). The following paragraphs outline the use of our independent (flood exposure) and dependent variables (environmental attitudes, environmental behaviors, support for climate policies and prosociality) for our analysis.

	data set 1 (Mar/Apr 2022)	data set 2 (Jul/Aug 2022)	data set 3 (Nov/Dec 2021)
	N = 6,285	N = 6,003	N = 1,466
Flood Exposure			
Living in an exposed region (by postal code)	\checkmark	\checkmark	\checkmark
Self-reported exposure	\checkmark		
Damage to persons or property in household	\checkmark		
Aid received during or after flood	\checkmark		
Environmental attitudes			
Impact of climate change	\checkmark		
Existence of climate change		\checkmark	
Climate change manmade		\checkmark	
Dangers of climate change		\checkmark	
Importance of climate policy		\checkmark	
Relative importance		\checkmark	
Seriousness of climate change			\checkmark
Importance of protecting the environment			\checkmark
Environmental behaviors			
Environmentally friendly behavior	\checkmark		
Green Party preference	\checkmark	\checkmark	
Conserving energy at home			\checkmark
Donation to atmosfair			\checkmark
Support for climate policies			
Increase in carbon pricing	\checkmark		
No new vehicles with combustion engines after 2035	\checkmark		
No new vehicles with combustion engines after 2030		\checkmark	
Constant increases in tax on petrol and diesel		\checkmark	
Speed limit of 130km/h on highways		\checkmark	
Coal phase-out			\checkmark
Expansion of electric vehicles			\checkmark
Expansion of public transport			\checkmark
Prosociality			
Positive reciprocity	\checkmark		
Negative reciprocity	\checkmark		
Trust	\checkmark		
Altruism	\checkmark		
Dictator game donation	\checkmark		
Past donations	\checkmark		
Past donations (amount)	\checkmark		
Volunteer work	\checkmark		
Socioeconomics			
Gender	√	✓	√
Age	√	√	✓
Income	√	✓	✓
Education	√	√	✓
Size of household	\checkmark	\checkmark	\checkmark

Table 1 Data sets and included variables

Note: Checkmarks indicate that individual-level information is available in the respective data set.

3.2 Measurement Variables

3.2.1 Measuring flood exposure

Flood exposure was measured by assigning participants' postal codes to exposed and nonexposed regions based on the flood mapping from the Federal Office for Civil Protection and Disaster Assistance (BBK) (Federal Ministry of the Interior and Federal Ministry of Finance, 2021). This allows us to compare relevant outcome variables across these two groups.

Figure 1 shows the exposed districts in Germany. The proportion of people living in exposed regions is similar across the three samples. Specifically, there are 18.30% living in regions exposed to the flood in data set 1 (n = 1,150 exposed and 5,135 not exposed), 19.32% in data set 2 (n = 1,160 exposed and 4,843 not exposed) and 19.58% in data set 3 (n = 287 exposed and 1,179 not exposed). Using external data on population size in each postal code area for a comparison, our exposure measure results in 16.198 million people living in exposed regions, which amounts to 19.20% of the general population in Germany (Statistisches Bundesamt [Destatis], 2023).





Note: Flood mapping of the July 2021 floods from the Federal Office for Civil Protection and Disaster Assistance (BBK). Orange color indicates exposure; red color indicates severe exposure (state of emergency) to the floods. Map outlines based on data from GeoBasis-DE (BKG, n.d.).

Without doubt, this exposure measure by the postal code liberally includes a relatively large group of people, many of whom have likely not been directly affected by the floods and have not personally suffered any substantial damage. The exposure measure must therefore be understood as either being directly affected or being in the proximity of the disaster, both of which can have an

impact on environmental and prosocial preferences. We go into more detail in the "Robustness Checks and Additional Analyses" subsection (4.5) by looking at respondents from areas that were severely exposed, i.e., participants from districts that were declared a state of emergency by the Federal Office for Civil Protection and Disaster Assistance (BBK) during the flood (see red colored area in Figure 1, Federal Ministry of the Interior and Federal Ministry of Finance 2021). These reduce the proportion of exposed participants in the sample to 3.21% in data set 1 (n = 202 exposed and 6,083 not exposed), 3.13% in data set 2 (n = 188 exposed and 5,815 not exposed) and 3.68% (n = 54 exposed and 1,412 not exposed). This is also close to the expected proportion when comparing to external data on populations in exposed regions, which is 2.94% (Statistisches Bundesamt [Destatis], 2023).

We also make use of additional, detailed self-reported information on flood exposure collected in the survey for data set 1. These are summarized into three relevant variables: First, whether respondents self-report their place of residence was affected by the flood. Second, whether any person from their household actually sustained any injuries or physical damage to their property and, third, whether any external aid, monetarily or in-kind, was provided to the household.

To rule out false attribution of potential effects to the flood exposure that may in fact be due to other, potentially pre-existing, differences across regions or samples, comparisons of socioeconomic characteristics of those exposed and not exposed are presented across all three data sets in the Supplementary Information (Table A.1). It shows that the groups are mostly balanced in their characteristics.

3.2.2 Measuring environmental attitudes

Regarding environmental attitudes, we measure (i) the perceived impact of climate change on one's personal living conditions, (ii) the participants' belief in the existence of climate change, (iii) whether they believe climate change is manmade, (iv) whether climate change is a danger to humankind, (v) whether they consider climate policy to be an important field in politics, (vi) its perceived importance compared to other issues, as well as (vii) the perceived seriousness of climate change as a problem, and (viii) an assessment of the importance of protecting the environment.

3.2.3 Measuring environmental behavior

Distinguished from attitudes, we also measure participants' pro-environmental behavior and include (i) self-assessed environmentally friendly behavior, as well as (ii & iii) the preference for

the German Green Party². Further, data set 3 entails whether respondents indicate to (iv) take actions to conserve energy at home and (v) their decision in an incentivized donation experiment. Within the survey, respondents were asked to allocate 100 Euros between themselves and their donation to *atmosfair*, a non-profit organization which uses donations for climate friendly projects to offset greenhouse gas emissions. An example was shown that describes the construction of micro power-grids and solar panels in villages in developing countries, which allow for clean rural electricity provision. Since not only the environment but also other people directly benefit from these donations, the experimental decision could similarly be interpreted as a measurement of prosocial behavior.³ Participants were informed that every 100th decision was implemented.

3.2.4 Measuring support for climate policies

To measure the support for climate policy measures, we ask whether respondents support (i) higher taxes on carbon dioxide emissions, (ii) bans on new vehicles with combustion engines after 2035 or (iii) 2030, (iv) constant increases in tax on petrol and diesel, (v) a general speed limit of 130km/h on highways, (vi) the phase-out of electricity production from burning coal, (vii) the expansion of electric vehicles and (viii) the expansion of public transport.

3.2.5 Measuring prosociality

Regarding prosociality, we only use the data from data set 1, which contains several complementary measures: four self-assessed items measuring general prosociality on 11-point response scales developed by Falk et al. (2018), that are (i) positive reciprocity, (ii) negative reciprocity, (iii) trust, and (iv) altruism. In a (v) dictator game experiment, participants were asked to allocate 10 tokens, each worth 10 Euros, between themselves and another, unknown survey participant⁴. The last block is about self-reported real-world behaviors, that include whether participants (vi) have donated money in the past year, and (vii) the respective amount if applicable. For the donation amount, we winsorize at the top 1% to account for potential outliers. Lastly, we elicit whether respondents (viii) regularly do volunteer work.

² Individuals who state to not have any political preference at all are considered to not have a preference for the Green party. Those who state that they do not wish to disclose their preference for a specific party are not considered for the analysis.

³ Yet, showing the same outcome twice would be redundant, thus, we decide to keep the results from the donation decision in the "environmental behavior" subsection.

⁴ In this survey, five participants were randomly selected whose decisions were then implemented. The allocated amounts were paid out via gift vouchers using "wunschgutschein.de" as a provider of vouchers. Participants were informed when playing the dictator game about the randomized selection of actual payouts.

3.3 Method of Analysis

For the main analysis presented in the following "Results" section (4), we show mean responses and corresponding standard deviations for metric and response-scale⁵ outcome variables across exposed and unexposed respondents for an intuitive comparison similar to Spence et al. (2011) and Lohmann and Kontoleon (2023). For dichotomous variables, we show proportions. Moreover, differences in means or proportions, respectively, between the two groups are shown and twosample t-tests are used for metric and response-scale outcome variables and Chi-squared tests are used for dichotomous ones to test for the significance of potential differences.

3.4 Robustness Checks and Detailed Analyses

We test the robustness of our results by applying a number of alternative and additional analyses. First, we run regression analyses in addition to the simple t- and chi2- tests that allow the inclusion of covariates. We rely on simple ordinary-least-squares (OLS) regressions, also in the case of dichotomous outcome variables. The latter are known as linear-probability models (LPM). We control for common socioeconomic characteristics, that are, gender, age, education, household income, and household size. Additionally, we add a dichotomous variable to control for being located in the eastern states of Germany.⁶ Second, data set 1 and data set 2 were established as panel data sets and thereby allow us to additionally make use of data available from previous waves conducted before the flood with, partly, the same participants. For a few variables, we can therefore not only measure cross-sectional differences but complement them by comparing changes over time, i.e., conducting a differences-in-differences analysis. The previous waves were collected in June 2019 (data set 1) and June 2021, shortly before the flood (data set 2). Finally, while we select the more general flood exposure, that is, living in the exposed areas, for the main analysis, we also look separately at those who are more severely exposed (declared state of emergency). Further, we go into more detail within data set 1 by using a self-reported affectedness by the floods as the relevant exposure measure as well as additional, detailed questions on damage suffered and aid received.

⁵ Most but not all response scales are Likert-scales.

⁶ Even 30 years after the reunification, cultural differences may be found in these five re-established states of the former German Democratic Republic that affect attitudes regarding the environment and prosociality.

3.5 Statistical Power for Analysis

Given the unexpected and unpredictable nature of the flood event, the sample sizes of exposed and unexposed respondents were determined by the proportion of affected respondents that each data set provides. With all three data sets being representative on national level and only some areas of Germany being exposed to the floods, the shares of those affected are consequently much smaller than the shares of those unaffected.⁷ For the power analysis, we therefore do not calculate a minimum sample size but base our number of observations of those exposed and not exposed on what we have actually observed in our data and report minimal detectable effect (MDE) sizes for each dependent variable. The MDEs are calculated based on the standard deviation (or proportions, in the case of dichotomous variables) for each variable observed in our data. The sizes of the presented MDEs (Supplementary Information Tables B.1 - B.4) give an intuitive interpretation of how large differences (i.e., "effects") would need to be in order to be considered statistically significant.⁸ It becomes evident that the MDEs do not appear excessively large for any of the tested variables.

4 Results

Firstly, results for the effect of being exposed to the floods on environmental attitudes (4.1) are presented, followed by its effects on environmental behavior (4.2) and on support for climate policies (4.3). We then look at prosociality (4.4) and add results from various robustness checks as well as additional analyses to the end of this section (4.5).

4.1 Environmental Attitudes

Table 2 compares the mean values of the survey items regarding environmental attitudes, such as the perception of the existence, causes and consequences of climate change, across those unexposed and exposed to the flood in July 2021. The last column of the table presents the differences in means and the p-values of the two-sided t-tests (if metric or response-scale) or chi-squared tests (if dichotomous). Results show no statistically significant difference between those

⁷ There may even be some unbalanced attrition in survey participation amongst those affected by the floods through still being, on average, busier with coping and reconstruction than those unaffected. However, in comparison to external data on population proportions exposed, our data does not support this conjecture, at least not on the level of allocation by postal codes.

⁸ Since results in section 4 will show no effects of flood exposure in the main analysis, a decision was made post-hoc to not make any multiple hypotheses adjustments to the statistical testing procedures despite the large range of indicators included. Hence, null effects, as found in all tested outcomes of the main analysis, can be considered even more robust.

not exposed and those exposed suggesting that experiencing the flood had no effect on environmental attitudes.

Table 2 Environmental Attitudes

			Not exposed	Exposed	
Survey	Question/Statement	Posponso ontions	means (std. dev.)		difference (p-value)
data set 1	Questionstatement	Response options	Ľ		
Impact of climate change	What do you think the consequences of climate change will be for your personal living conditions in the coming decades?	Five-point scale (very negative – very positive consequences)	3.793 (0.701) [4,966]	3.781 (0.717) [1,110]	-0.012 (0.598)
data set 2					
Existence of climate change	Based on everything you know; do you think the world's climate is changing or not?	Yes/No	0.928 - [4,781]	0.923	-0.005 (0.538) ^C
Climate change manmade	Do you think climate change is mostly caused by humans (as opposed to mostly caused by natural factors)?	Yes/No	0.884 [4,764]	0.868	-0.016 (0.123) ^C
Dangers of climate change	Do you think climate change is dangerous for humanity?	Four-point scale (certainly not – certainly)	3.610 (0.666) [4,427]	3.593 (0.671) [1,056]	-0.017 (0.448)
Importance of climate policy	Do you consider climate policy to be an important field in politics?	Four-point scale (not important – important)	3.667 (0.723) [4,823]	3.628 (0.783) [1,153]	-0.039 (0.101)
Relative importance	In your opinion, are the environment and climate change one of the most important issues Germany is currently facing?	Yes/No	0.446 - [4,843]	0.435 - [1,160]	-0.010 (0.520) ^C
data set 3					
Seriousness of climate change	How serious a problem do you think climate change is at this moment?	Ten-point scale (not at all a serious problem – an extremely serious problem)	7.463 (2.388) [1,179]	7.206 (2.493) [287]	-0.258 (0.105)
Importance of protecting the environment	How important is protecting the environment to you personally?	Four-point scale (not at all important – very important)	3.259 (0.670) [1,179]	3.188 (0.631) [287]	-0.071 (0.106)

For the calculation of means and proportions, "Do not know" and "Prefer not to say" responses were omitted. Two-sided t-tests for metric and response-scale variables

C = Chi-squared test if variable is dichotomous Significance levels: * p < 0.10, ** p < 0.05, *** p < 0.01

4.2 Environmental Behavior

Table 3 presents comparisons of environmental behaviors, including self-assessed environmental action, preferences for the German Green Party and results from an incentivized donation experiment that benefits a pro-environmental organization. Again, we find no statistically significant differences in means across those not exposed and those exposed to the floods.

Table 3 Environmental Behavior

			Not exposed	Exposed	
Survey	Question/Statement	Response options	means (std. dev.)		difference (p-value)
data set 1					
Environmentally friendly behavior	I behave in an environmentally conscious manner, even if this is associated with higher costs and/or efforts.	Five-point scale (strongly disagree – strongly agree)	3.333 (0.780) [5,082]	3.325 (0.775) [1,140]	-0.008 (0.757)
Green Party preference	If you lean towards a certain party, do you have a preference for the Green Party?	Yes/No	0.187 - [4,819]	0.172 [1,080]	-0.015 (0.259) ^C
data set 2					
Green Party preference	If you lean towards a certain party, do you have a preference for the Green Party?	Yes/No	0.238 [4,639]	0.218	-0.020 (0.155) ^C
data set 3					
Conserving energy at home	How much do you agree or disagree with the following statements? "I take action to conserve energy at home"	Five-point scale (strongly disagree – strongly agree)	4.002 (0.837) [1,179]	4.042 (0.852) [287]	-0.040 (0.468)
Donation to atmosfair	Donation Experiment: How much of 100 € would you like to donate to atmosfair?	Values between 0 and 100	38.686 (30.694) [1,179]	37.265 (31.444) [287]	-1.421 (0.484)

For the calculation of means and proportions, "Do not know" and "Prefer not to say" responses were omitted.

Two-sided t-tests for metric and response-scale variables

C = Chi-squared test if variable is dichotomous

Significance levels: * p < 0.10, ** p < 0.05, *** p <0.01

4.3 Support for Climate Policies

Table 4 examines differences in the acceptance of environmental policies, specifically policies that aim to decrease carbon dioxide emissions. Differences for most variables are not significant. For two measures – support for coal phase-out and the expansion of public transport – we observe a small but statistically significant decrease in average support among those who live in regions exposed to the flood. However, these differences are relatively small and do not appear to indicate a broad trend.

			Not exposed	Exposed	
Survey	Ouestion/Statement	Response options	means (std. dev.)		difference (p-value)
data set 1	x				
Increase in carbon pricing	What is your stance on the following measure?	Five-point scale (strongly disagree – strongly agree)	2.845 (1.315) [3,958]	2.810 (1.302) [923]	-0.034 (0.473)
No new vehicles with combustion engines after 2035	What is your stance on the following measure?	Five-point scale (strongly disagree – strongly agree)	2.628 (1.384) [4,990]	2.588 (1.371) [1,129]	-0.040 (0.378)
data set 2					
No new vehicles with combustion engines after 2030 ^A	Do you support the following measure?	Five-point scale (No – Yes)	2.599 (1.470) [4,755]	2.577 (1.426) [1,144]	-0.022 (0.651)
Constant increase in tax on petrol and diesel ^A	Do you support the following measure?	Five-point scale (No – Yes)	2.186 (1.296) [4,715]	2.137 (1.235) [1,132]	-0.049 (0.250)
Speed limit of 130km/h on highways ^A	Do you support the following measure?	Five-point scale (No – Yes)	3.772 (1.553) [4,817]	3.773 (1.544) [1,154]	-0.001 (0.983)
data set 3					
Coal phase-out	Please indicate what your opinion is on the following policy measure.	Five-point scale (strongly disagree – strongly agree)	3.696 (1.159) [1,179]	3.568 (1.135) [287]	-0.128* (0.091)
Expansion of electric vehicles	Please indicate what your opinion is on the following policy measure.	Five-point scale (strongly disagree – strongly agree)	3.221 (1.278) [1,179]	3.146 (1.306) [287]	-0.075 (0.375)
Expansion of public transport	Please indicate what your opinion is on the following policy measure.	Five-point scale (strongly disagree – strongly agree)	4.087 (0.913) [1,179]	3.965 (0.978) [287]	-0.121** (0.047)

Table 4 Support for Climate Policies

For the calculation of means and proportions, "Do not know" and "Prefer not to say" responses were omitted.

Two-sided t-tests for metric and response-scale variables

A = The question was preceded by an introductory statement: "For the mobility transition, there are some measures being discussed that aim to reduce car use and thereby emission of pollutants and carbon dioxide."

Significance levels: * p < 0.10, ** p < 0.05, *** p <0.01

4.4 Prosociality

Table 5 reports mean values for prosociality measures, including self-assessed altruism, reciprocity, and trust, as well as decisions in an incentivized donation experiment and a set of real-world behaviors. This analysis is based on data set 1 only. Of all four hypothetical, self-assessed measures, only altruism, evaluated by the hypothetical willingness to give to good causes, shows a statistically significant higher mean value for those exposed to the flood (diff. = 0.162, p = 0.049). However, the magnitude of this difference on an eleven-point scale is modest. Other measures, such as positive or negative reciprocity, trust, and decisions in an incentivized donation experiment, show no significant differences. In line with these findings, reported actual behaviors such as volunteer work and the amount of money donated to charity are not affected by flood exposure. Lastly, while a slightly higher proportion of exposed respondents reported donating money in the past year (diff = 2.2 percentage points, p = 0.087), the difference is only marginally significant.

			Not exposed	Exposed	
Survey	Question/Statement	Response options	means (std. dev.) [n]		difference (p-value)
data set 1					
Positive reciprocity	Willingness to return a favor	Eleven-point scale	9.364 (1.740) [5,027]	9.395 (1.637) [1,128]	-0.031 (0.586)
Negative reciprocity	Willingness to take revenge	Eleven-point scale	3.563 (2.455) [4,982]	3.590 (2.394) [1,117]	-0.027 (0.739)
Trust	"People have only the best intentions"	Eleven-point scale	6.264 (2.347) [4,967]	6.209 (2.299) [1,118]	-0.054 (0.484)
Altruism	Willingness to give to good causes	Eleven-point scale	7.916 (2.498) [5,008]	8.078 (2.456) [1,123]	-0.162** (0.049)
Dictator game donation	Indicate how many tokens you would like to give to your unknown partner	Tokens between 0 and 10	5.160 (1.464) [3,669]	5.196 (1.504) [869]	-0.036 (0.518)
Past donations	Did you donate money last year, that is, in 2021?	Yes/No	0.800 - [4,961]	0.823	-0.022* (0.087) ^C
Past donations (amount)	What was the total amount you donated last year?	Values ≥ 0	311.022 (542.427) [4,411]	328.640 (530.877) [980]	-17.617 (0.356)
Volunteer work	Do you regularly engage in voluntary work?	Yes/No	0.428	0.415	-0.013 (0.411) ^C

Table 5 Prosociality

For the calculation of means and proportions, "Do not know" and "Prefer not to say" responses were omitted. Two-sided t-tests for metric and response-scale variables

C = Chi-squared test if variable is dichotomous

Significance levels: * p < 0.10, ** p < 0.05, *** p < 0.01

4.5 Robustness Checks and Detailed Analyses

The absence of significant effects of flood exposure on environmental attitudes and behaviors is confirmed by regression analyses that adjust for covariates (Supplementary Information Tables C.1 - C.3) as well as the difference-in-differences estimations, which compare changes over time (Supplementary Information Tables D.1 – D.2). When using the alternative allocation of exposure by focusing on severely exposed respondents, results do not change, either (Supplementary Information Tables E.1 – E.3). Also concerning prosociality, null results are confirmed when including covariates or using the severe exposure measure (Supplementary Information Tables C.4 and E.4).

Within the survey that was conducted for data set 1, participants were asked whether their place of residence had been affected by the flood in July 2021. In case of reported exposure, more detailed information was collected on the flood experiences. Of the whole nationwide sample, 218 (3.48%) respondents reported being affected. Responses to flood-related questions were used to create additional subgroups. We first distinguish whether someone from the respondent's household suffered any injuries to persons from their household and/or physical damage to their property. A total of 70 (1.12%) persons within our data set belong to this group. Next, another distinction is made within those who sustained damage by whether respondents indicated having received any form of aid, either monetarily or in-kind. There are 40 (0.64%) respondents in this group. Given these rather small sample sizes, we focus for the analysis within this subsection on the most striking findings and interpret these with caution.

For environmental attitudes and behaviors, none of these distinctions result in any meaningful differences, indicating that not even suffering physical damage has had any impact on environmental attitudes and behaviors (see Supplementary Information Table F.1 for means and proportions and Table F.2 for test for significance of differences).

For prosociality, however, we observe significant differences in some outcome variables when comparing varying degrees of affectedness (Table 6 for means and proportions, Supplementary Information Table F.3 for test for significance of differences). Specifically, respondents who indicated that their place of residence had been affected by the flood (n = 218) score lower on self-assessed positive reciprocity (two-sided t-test for (1) vs. (2): diff. = -0.359, p-value = 0.003), regardless of whether they also report having sustained any damage. For self-reported negative reciprocity, the same group shows higher average scores, and even higher scores when they have suffered damage (two-sided t-test for (1) vs. (2): diff. = 0.383, p-value = 0.003; for (3) vs. (4): diff. = 0.792, p-value = 0.007). They further show significantly higher scores in self-assessed altruism

(two-sided t-test for (1) vs. (2): diff. = 0.363, p-value = 0.036), while there are no differences in the average scores for self-assessed trust.

	(1)	(2)	(3)	(4)	(5)	(6)
	Not affected			Affected		
Affectedness		All	No damage	Damage	Damage but no aid received	Damage and aid received
Self-reported exposure:	no	yes	yes	yes	yes	yes
Damage:			no	yes	yes	yes
Aid:					no	yes
Prosociality variable						
Positive reciprocity	9.382	9.023	9.048	8.971	8.433	9.375
	(1.712)	(1.952)	(1.852)	(2.160)	(2.459)	(1.835)
Negative reciprocity	3.556	3.940	3.747	4.348	4.600	4.154
	(2.437)	(2.594)	(2.483)	(2.791)	(2.872)	(2.749)
Trust	6.257	6.160	6.238	6.000	6.333	5.744
	(2.341)	(2.278)	(2.233)	(2.376)	(2.644)	(2.149)
Altruism	7.935	8.298	8.466	7.942	7.433	8.333
	(2.489)	(2.503)	(2.506)	(2.479)	(2.622)	(2.321)
Dictator game donation	5.164	5.232	5.281	5.130	5.190	5.091
	(1.470)	(1.500)	(1.605)	(1.260)	(1.940)	(0.522)
Past donations (d)	0.805	0.808	0.814	0.794	0.733	0.842
	-	-	-	-	-	-
Past donations (amount)	311.882	384.160	435.674	271.525	324.000	232.941
	(539.177)	(574.266)	(609.281)	(474.555)	(664.467)	(266.623)
Volunteering (d)	0.424	0.493	0.503	0.471	0.367	0.553
- · ·	-	-	-	-	-	-
Observations	6,043	218	148	70	30	40

Table 6 Degrees (f self-reported affectedness (on prosociality
-------------------	--------------------------------	-----------------

Means shown for metric and response-scale variables with standard deviations in parentheses, proportions shown for dichotomous variables. When comparing means, note that groups (3) and (4) are subgroups of group (2). and groups (5) and (6) are subgroups of group (4). (d) denotes dichotomous variables.

Although average donations in the incentivized dictator game do not differ across all groups, the average amounts spent on real-world charities among those who donated are higher among respondents whose place of residence was affected but whose household did not sustain any damage (two-sided t-test for (1) vs. (3): diff. = 123.79 Eur, p-value = 0.010), while donations are lower among respondents who experienced damage (two-sided t-test for (1) vs. (4): diff. = -40.36 Eur, p-value = 0.567; for (3) vs. (4): diff. = -164.15 Eur, p-value = 0.069, see Supplementary Information Table F.3).

When comparing individuals who received some form of aid to those who did not, the sample size becomes even smaller, and differences rarely reach statistical significance. However, those who received aid tend to report higher scores in self-assessed positive reciprocity and altruism. Recipients of aid also show higher probabilities of engaging in volunteer work. Regarding involvement in regular volunteer work, there seems to be a general positive association with self-reported exposure to the flood.

5 Discussion

Exploring the relationship between experiencing natural disasters and environmental attitudes and behaviors has produced mixed results in the literature (Howe et al., 2019). Our study contributes to the ongoing debate by investigating the impact of the July 2021 flood disaster in Germany. Using three distinct nationwide and representative survey data sets, we assessed a wide range of individual-level indicators on environmental attitudes, environment-related behaviors, and support for climate policies. Despite the comprehensiveness of our approach, we found no significant effects on any of our selected outcome measures.

When comparing our results to existing studies, some differences emerge. Research on previous disasters has identified links between exposure and risk perceptions (Frondel et al., 2017), increased belief in climate change (Osberghaus & Fugger, 2022) and stronger pro-environmental attitudes (Bulut & Samuel, 2024). However, our findings contrast those and instead align better with recent studies focusing on the same 2021 flood event in Germany. Studies investigating Green Party support in flood-affected regions during the 2021 federal elections observed only marginal increases, which dissipated within weeks (Garside & Zhai, 2022; Hilbig & Riaz, 2024). Our results, reflecting medium- to long-term effects collected months to a year after the event, similarly indicate no lasting impacts.

A critical factor identified in the literature is whether exposure includes proximity to the disaster or personal damage (Frondel et al., 2017; Holub & Schündeln, 2023; Osberghaus & Demski, 2019). In our study, alternative specifications accounting for varying severity of exposure – including proximity to the event and personal damage – yielded no significant effects. It has been argued that the intensive media coverage of the flood may have diluted potential localized effects, influencing attitudes across the entire country rather than solely among those in exposed areas (Hilbig & Riaz, 2024; Sisco, 2021). However, our panel data analyses show no general, nationwide increases in environmental attitudes from pre- to post-flood periods (Supplementary Materials D.1 and D.2)

In the second part of this study, we analyzed the flood's impact on prosociality – an area often studied in the context of emerging economies where informal social networks play a critical role in disaster response. This is, to the best of our knowledge, the first study to explore how the 2021 flood's effects on prosociality in Germany, reflecting an industrialized country setting. Despite analyzing several markers, including an incentivized donation experiment, we only found limited effects of flood exposure on prosocial outcomes.

Importantly, only with self-reported affectedness as the explanatory variable, some notable variation emerged. Affected respondents reported lower scores in positive reciprocity and higher scores in negative reciprocity as well as in altruism. These patterns suggest that affected individuals prioritize punitive responses to perceived negative behavior over affirming positive behavior. This tendency could reflect efforts to regain a sense of control or to address frustrations linked to their experiences. At the same time, increased altruism may reflect heightened empathy and solidarity driven shared sense of suffering.

Consistent with the findings on altruism, affected individuals reported higher levels of past realworld donations than non-exposed respondents, but only so when they have not also sustained personal damage to their household by the flood. Drawing on the potential mechanisms presented in section 1, this suggests that proximity to the disaster – potentially including witnessing spatially and socially close others suffer – can increase empathy, thereby motivating prosocial behaviors (Vardy & Atkinson, 2019). In contrast, we observed lower past real-world donations among respondents who stated to have sustained damage from the flood, even when compared to nonaffected respondents. Individuals who experienced personal damage are likely preoccupied with coping, reconstructions efforts and meeting their own financial needs to recover from the flood (Osberghaus & Demski, 2019). It seems intuitive that victims of disasters do not necessarily prioritize donating to their fellow victims, while those living in exposed areas but have not suffered any damage are more likely to have the resources and time to engage in prosocial activities.

A closer differentiation among participants who suffered personal damage revealed differences between those who had received any form of external aid and those who had not. Respondents who received aid exhibited elevated scores for certain markers of prosociality, including positive reciprocity, compared to those who suffered damage but did not receive any aid. These findings align with existing literature and the mechanisms previously identified (Andrabi & Das, 2017; Becchetti et al., 2017; Cassar et al., 2017; Rahman et al., 2020). We further found that self-reported exposure went along with a somewhat higher probability of doing volunteer work, but it did not seem to depend on whether the household had sustained personal damage or received external aid. Overall, many of the discussed differences turned out as statistically significant despite the focus on small subsamples for this part of the analysis.

6 Conclusion

Anticipating progressing climate change and an increase in the prevalence of natural disasters, it seems essential to study how people react to experiencing climate change impacts, such as flood disasters, not only in the direct aftermath but also beyond, that is, months and years after the event. Will increases in frequency, scale as well as intensity of disasters lead people to change their behavior towards more sustainable lifestyles and practices that contribute to mitigating the causes for climate change? Drawing on results from the three nationwide data sets, which were collected between 4 and 12 months after the 2021 flood in Germany, we found no evidence for changes in individual attitudes or behaviors regarding environment or climate change. This suggests that even the first-hand experience of natural disasters does not automatically drive individuals to take greater action against global warming. This apparent lack or deficiency of voluntary mitigation efforts may highlight the need for policy interventions that regulate harmful behaviors or incentivize sustainable ones. However, the feasibility and effectiveness of such policies depends to a large extent also on their public acceptance. We have, therefore, in our analysis additionally examined the support for a set of climate-related policy measures. Similar to environmental attitudes and behaviors, though, we could not find any increases in support for such measures among residents of regions exposed to the flood. Notably, environmental attitudes in our sample were already high, often in the upper quartiles of our scales of measurement. Apparently, though, these attitudes do not translate well into pro-environmental behaviors and support for climate policies, as there is, according to our data and type of measurement, still significant potential for improvement.

These findings suggest that we cannot assume increased proximity to natural disasters to inherently lead to greater awareness and climate action. Future research should explore how different ways of processing and coping with natural disaster shape individual responses, such as in environmental awareness and action. Key areas of exploration include the role of media framings, the attribution of disasters to climate change and the influence of motivated reasoning – a process in which individuals interpret information in ways that align with their existing beliefs. These factors could provide valuable insights into the drivers and barriers to environmental engagement.

Regarding prosociality, we observed no meaningful differences when comparing by disaster exposure on the municipality level. However, more detailed analyses revealed that those reporting to live in affected places without damage tended to donate more, whereas those who suffered damage donated less. This underscores the importance of distinguishing between the mere proximity to a disaster and experiencing personal damage. It seems intuitive that those who are closer to the disaster, both spatially and socially, are more motivated to help than more distant ones, while those whose households were personally affected are focused on their own recovery.

The flood's impact on prosociality appears most evident in real-world behaviors, such as monetary donations and volunteering. Given our data, we cannot determine whether these effects reflect actions tied to the particular flood event, namely donating money to flood victims and volunteering in clean up and reconstruction, or if it extends to more general prosocial behaviors unrelated to the disaster. In contrast, measures capturing broader and context-independent prosociality, i.e., self-reported hypothetical prosociality and incentivized donations in the dictator game, measured several months after the floods, showed little or less clear influence by the flood. Future research should distinguish between event-related and general prosociality in addition to considering the differing effects of witnessing a disaster due to spatial proximity versus experiencing personal damage.

*

References

- Abatayo, A. L., & Lynham, J. (2020). Risk preferences after a typhoon: An artefactual field experiment with fishers in the Philippines. *Journal of Economic Psychology*, 79, 102195. https://doi.org/10.1016/j.joep.2019.102195
- Ahmad, N., & Younas, M. Z. (2021). A blessing in disguise?: Assessing the impact of 2010–2011 floods on trust in Pakistan*. *Environmental Science and Pollution Research*, 28(20), 25419–25431. https://doi.org/10.1007/s11356-020-12332-4
- Ahsan, D. A. (2014). Does natural disaster influence people's risk preference and trust? An experiment from cyclone prone coast of Bangladesh. *International Journal of Disaster Risk Reduction*, 9, 48– 57. https://doi.org/10.1016/j.ijdrr.2014.02.005
- Andrabi, T., & Das, J. (2017). In Aid We Trust: Hearts and Minds and the Pakistan Earthquake of 2005. *The Review of Economics and Statistics*, 99(3), 371–386. https://doi.org/10.1162/REST_a_00638
- Apel, H., Vorogushyn, S., & Merz, B. (2022). Brief communication: Impact forecasting could substantially improve the emergency management of deadly floods: case study July 2021 floods in Germany. *Natural Hazards and Earth System Sciences*, 22(9), 3005–3014. https://doi.org/10.5194/nhess-22-3005-2022
- Baccini, L., & Leemann, L. (2021). Do natural disasters help the environment? How voters respond and what that means. *Political Science Research and Methods*, 9(3), 468–484. https://doi.org/10.1017/psrm.2020.25
- Bai, Y., & Li, Y. (2021). More suffering, more involvement? The causal effects of seismic disasters on social capital. World Development, 138, 105221. https://doi.org/10.1016/j.worlddev.2020.105221
- Becchetti, L., Castriota, S., & Conzo, P. (2017). Disaster, Aid, and Preferences: The Long-run Impact of the Tsunami on Giving in Sri Lanka. World Development, 94, 157–173. https://doi.org/10.1016/j.worlddev.2016.12.014
- Biener, C., & Landmann, A. (2023). Recovery mode: Non-cognitive skills after the storm. *World Development*, 164, 106152. https://doi.org/10.1016/j.worlddev.2022.106152
- BKG. (n.d.). *BKG Bundesamt für Kartographie und Geodäsie*. Home. Retrieved February 17, 2025, from https://www.bkg.bund.de/DE/Home/home.html
- Brief, A. P., & Motowidlo, S. J. (1986). Prosocial Organizational Behaviors. *The Academy of Management Review*, 11(4), 710. https://doi.org/10.2307/258391
- Bulut, H., & Samuel, R. (2024). The influence of the 2021 European flooding on pro-environmental attitudes and partial behaviour transition. *Npj Climate Action*, *3*(1), 22. https://doi.org/10.1038/s44168-024-00103-7
- Calo-Blanco, A., Kovářík, J., Mengel, F., & Romero, J. G. (2017). Natural disasters and indicators of social cohesion. *PLOS ONE*, *12*(6), e0176885. https://doi.org/10.1371/journal.pone.0176885
- Cassar, A., Healy, A., & von Kessler, C. (2017). Trust, Risk, and Time Preferences After a Natural Disaster: Experimental Evidence from Thailand. *World Development*, 94, 90–105. https://doi.org/10.1016/j.worlddev.2016.12.042
- Chantarat, S., Oum, S., Samphantharak, K., & Sann, V. (2019). Natural Disasters, Preferences, and Behaviors: Evidence from the 2011 Mega Flood in Cambodia. *Journal of Asian Economics*, 63, 44–74. https://doi.org/10.1016/j.asieco.2019.05.001
- Cornwall, W. (2021). Europe's deadly floods leave scientists stunned. *Science*, 373(6553), 372–373. https://doi.org/10.1126/science.373.6553.372
- CRED, U. (2020). Human Cost of Disasters. An Overview of the last 20 years: 2000–2019. CRED, UNDRR, Geneva, 609.
- Decety, J., Bartal, I. B.-A., Uzefovsky, F., & Knafo-Noam, A. (2016). Empathy as a driver of prosocial behaviour: Highly conserved neurobehavioural mechanisms across species. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 371(1686), 20150077. https://doi.org/10.1098/rstb.2015.0077

- Demski, C., Capstick, S., Pidgeon, N., Sposato, R. G., & Spence, A. (2017). Experience of extreme weather affects climate change mitigation and adaptation responses. *Climatic Change*, *140*(2), 149–164. https://doi.org/10.1007/s10584-016-1837-4
- Depow, G. J., Francis, Z., & Inzlicht, M. (2021). The Experience of Empathy in Everyday Life. *Psychological Science*, 32(8), 1198–1213. https://doi.org/10.1177/0956797621995202
- Deutsches Zentralinstitut für soziale Fragen. (2022). Ergebnis der DZI-Umfrage. Ein Jahr danach: Hochwasser-Spenden 2021. Deutsches Zentralinstitut für soziale Fragen. https://www.dzi.de/pressemitteilungen/ein-jahr-danach-655-millionen-euro-hochwasser-spenden/
- Dietze, M., Bell, R., Ozturk, U., Cook, K. L., Andermann, C., Beer, A. R., Damm, B., Lucia, A., Fauer, F. S., Nissen, K. M., Sieg, T., & Thieken, A. H. (2022). More than heavy rain turning into fast-flowing water a landscape perspective on the 2021 Eifel floods. *Natural Hazards and Earth System Sciences*, 22(6), 1845–1856. https://doi.org/10.5194/nhess-22-1845-2022
- Duijndam, S. J., Botzen, W. J. W., Endendijk, T., De Moel, H., Slager, K., & Aerts, J. C. J. H. (2023). A look into our future under climate change? Adaptation and migration intentions following extreme flooding in the Netherlands. *International Journal of Disaster Risk Reduction*, 95, 103840. https://doi.org/10.1016/j.ijdtr.2023.103840
- Dussaillant, F., & Guzmán, E. (2014). Trust via disasters: The case of Chile's 2010 earthquake. *Disasters*, 38(4), 808–832. https://doi.org/10.1111/disa.12077
- Eisenberg, N., & Miller, P. A. (1987). The relation of empathy to prosocial and related behaviors. *Psychological Bulletin*, 101(1), 91–119. https://doi.org/10.1037/0033-2909.101.1.91
- European Environment Agency. (2024, October 14). *Economic losses from weather- and climate-related extremes in Europe*. EEA. https://www.eea.europa.eu/en/analysis/indicators/economic-losses-from-climate-related
- Falk, A., Becker, A., Dohmen, T., Enke, B., Huffman, D., & Sunde, U. (2018). Global Evidence on Economic Preferences*. *The Quarterly Journal of Economics*, 133(4), 1645–1692. https://doi.org/10.1093/qje/qjy013
- Federal Ministry of the Interior and Federal Ministry of Finance. (2021). Zwischenbericht zur Flutkatastrophe 2021: Katastrophenhilfe, Soforthilfen und Wiederaufbau. https://www.bmi.bund.de/SharedDocs/kurzmeldungen/DE/2021/09/zwischenbericht-Flutkatastrophe.html
- Fekete, A., & Sandholz, S. (2021). Here Comes the Flood, but Not Failure? Lessons to Learn after the Heavy Rain and Pluvial Floods in Germany 2021. Water, 13(21), 3016. https://doi.org/10.3390/w13213016
- Fiala, O. (2017). Natural Disasters and Individual Behaviour in Developing Countries. Springer International Publishing. https://doi.org/10.1007/978-3-319-53904-1
- Fleming, D. A., Chong, A., & Bejarano, H. D. (2014). Trust and Reciprocity in the Aftermath of Natural Disasters. *The Journal of Development Studies*, 50(11), 1482–1493. https://doi.org/10.1080/00220388.2014.936395
- Frondel, M., Simora, M., & Sommer, S. (2017). Risk Perception of Climate Change: Empirical Evidence for Germany. *Ecological Economics*, *137*, 173–183. https://doi.org/10.1016/j.ecolecon.2017.02.019
- Garside, S., & Zhai, H. (2022). If not now, when? Climate disaster and the Green vote following the 2021 Germany floods. *Research & Politics*, 9(4), 20531680221141523. https://doi.org/10.1177/20531680221141523
- Haden, V. R., Niles, M. T., Lubell, M., Perlman, J., & Jackson, L. E. (2012). Global and Local Concerns: What Attitudes and Beliefs Motivate Farmers to Mitigate and Adapt to Climate Change? *PLoS ONE*, 7(12), e52882. https://doi.org/10.1371/journal.pone.0052882
- Hilbig, H., & Riaz, S. (2024). Natural Disasters and Green Party Support. *The Journal of Politics*, 86(1), 241–256. https://doi.org/10.1086/726917

- Hirabayashi, Y., Mahendran, R., Koirala, S., Konoshima, L., Yamazaki, D., Watanabe, S., Kim, H., & Kanae, S. (2013). Global flood risk under climate change. *Nature Climate Change*, 3(9), 816–821. https://doi.org/10.1038/nclimate1911
- Hoffmann, R., Muttarak, R., Peisker, J., & Stanig, P. (2022). Climate change experiences raise environmental concerns and promote Green voting. *Nature Climate Change*, *12*(2), 148–155. https://doi.org/10.1038/s41558-021-01263-8
- Holub, F., & Schündeln, M. (2023). Pro-environmental voting when climate change is made salient: Evidence from high-resolution flooding data. *PLOS Climate*, 2(8), e0000219. https://doi.org/10.1371/journal.pclm.0000219
- Hornsey, M. J., Harris, E. A., Bain, P. G., & Fielding, K. S. (2016). Meta-analyses of the determinants and outcomes of belief in climate change. *Nature Climate Change*, 6(6), 622–626. https://doi.org/10.1038/nclimate2943
- Howe, P. D., Marlon, J. R., Mildenberger, M., & Shield, B. S. (2019). How will climate change shape climate opinion? *Environmental Research Letters*, 14(11), 113001. https://doi.org/10.1088/1748-9326/ab466a
- IPCC. (2021). Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press.
- Klöckner, C. A. (2013). A comprehensive model of the psychology of environmental behaviour—A metaanalysis. *Global Environmental Change*, 23(5), 1028–1038. https://doi.org/10.1016/j.gloenvcha.2013.05.014
- Kuroishi, Y., & Sawada, Y. (2019). Motivations behind prosocial behavior: Evidence from the Philippines. *Journal of Asian Economics*, 64, 101127. https://doi.org/10.1016/j.asieco.2019.06.003
- Lee, T. M., Markowitz, E. M., Howe, P. D., Ko, C.-Y., & Leiserowitz, A. A. (2015). Predictors of public climate change awareness and risk perception around the world. *Nature Climate Change*, 5(11), 1014–1020. https://doi.org/10.1038/nclimate2728
- Lehmann, K., Böckler, A., Klimecki, O., Müller-Liebmann, C., & Kanske, P. (2022). Empathy and correct mental state inferences both promote prosociality. *Scientific Reports*, 12(1), 16979. https://doi.org/10.1038/s41598-022-20855-8
- Lim, D., & DeSteno, D. (2016). Suffering and compassion: The links among adverse life experiences, empathy, compassion, and prosocial behavior. *Emotion*, *16*(2), 175–182. https://doi.org/10.1037/emo0000144
- Lohmann, P. M., & Kontoleon, A. (2023). Do Flood and Heatwave Experiences Shape Climate Opinion? Causal Evidence from Flooding and Heatwaves in England and Wales. *Environmental and Resource Economics*, 86(1–2), 263–304. https://doi.org/10.1007/s10640-023-00796-0
- Lorenzoni, I., & Pidgeon, N. F. (2006). Public Views on Climate Change: European and USA Perspectives. *Climatic Change*, 77(1–2), 73–95. https://doi.org/10.1007/s10584-006-9072-z
- Maki, A., Dwyer, P. C., Blazek, S., Snyder, M., González, R., & Lay, S. (2019). Responding to natural disasters: Examining identity and prosociality in the context of a major earthquake. *British Journal* of Social Psychology, 58(1), 66–87. https://doi.org/10.1111/bjso.12281
- Marx, S. M., Weber, E. U., Orlove, B. S., Leiserowitz, A., Krantz, D. H., Roncoli, C., & Phillips, J. (2007). Communication and mental processes: Experiential and analytic processing of uncertain climate information. *Global Environmental Change*, 17(1), 47–58. https://doi.org/10.1016/j.gloenvcha.2006.10.004
- McDonald, R. I., Chai, H. Y., & Newell, B. R. (2015). Personal experience and the 'psychological distance' of climate change: An integrative review. *Journal of Environmental Psychology*, 44, 109–118. https://doi.org/10.1016/j.jenvp.2015.10.003
- Méon, P.-G., & Verwimp, P. (2022). Pro-social behavior after a disaster: Evidence from a storm hitting an open-air festival. *Journal of Economic Behavior & Organization*, 198, 493–510. https://doi.org/10.1016/j.jebo.2022.04.010

- Mewes, L., Tuitjer, L., & Dirksmeier, P. (2024). Exploring the variances of climate change opinions in Germany at a fine-grained local scale. *Nature Communications*, 15(1), 1867. https://doi.org/10.1038/s41467-024-45930-8
- Netzel, L. M., Heldt, S., Engler, S., & Denecke, M. (2021). The importance of public risk perception for the effective management of pluvial floods in urban areas: A case study from Germany. *Journal of Flood Risk Management*, *14*(2), e12688. https://doi.org/10.1111/jfr3.12688
- Ogunbode, C. A., Demski, C., Capstick, S. B., & Sposato, R. G. (2019). Attribution matters: Revisiting the link between extreme weather experience and climate change mitigation responses. *Global Environmental Change*, *54*, 31–39. https://doi.org/10.1016/j.gloenvcha.2018.11.005
- Ogunbode, C. A., Doran, R., Hanss, D., Ojala, M., Salmela-Aro, K., Van Den Broek, K. L., Bhullar, N., Aquino, S. D., Marot, T., Schermer, J. A., Wlodarczyk, A., Lu, S., Jiang, F., Maran, D. A., Yadav, R., Ardi, R., Chegeni, R., Ghanbarian, E., Zand, S., ... Karasu, M. (2022). Climate anxiety, wellbeing and pro-environmental action: Correlates of negative emotional responses to climate change in 32 countries. *Journal of Environmental Psychology*, 84, 101887. https://doi.org/10.1016/j.jenvp.2022.101887
- Osberghaus, D., & Demski, C. (2019). The causal effect of flood experience on climate engagement: Evidence from search requests for green electricity. *Climatic Change*, 156(1–2), 191–207. https://doi.org/10.1007/s10584-019-02468-9
- Osberghaus, D., & Fugger, C. (2022). Natural disasters and climate change beliefs: The role of distance and prior beliefs. *Global Environmental Change*, 74, 102515. https://doi.org/10.1016/j.gloenvcha.2022.102515
- Peisker, J. (2023). Context matters: The drivers of environmental concern in European regions. *Global Environmental Change*, 79, 102636. https://doi.org/10.1016/j.gloenvcha.2023.102636
- Rahman, M. H., Lee, G. H. Y., Shabnam, N., & Jayasinghe, S. (2020). Weathering trust. *Journal of Economic Behavior & Organization*, 178, 449–473. https://doi.org/10.1016/j.jebo.2020.07.027
- Reichstein, M., Riede, F., & Frank, D. (2021). More floods, fires and cyclones—Plan for domino effects on sustainability goals. *Nature*, 592(7854), 347–349. https://doi.org/10.1038/d41586-021-00927-x
- Rodríguez, H., Trainor, J., & Quarantelli, E. L. (2006). Rising to the Challenges of a Catastrophe: The Emergent and Prosocial Behavior following Hurricane Katrina. *The ANNALS of the American Academy of Political and Social Science*, 604(1), 82–101. https://doi.org/10.1177/0002716205284677
- Rüttenauer, T. (2024). More talk, no action? The link between exposure to extreme weather events, climate change belief and pro-environmental behaviour. *European Societies*, 26(4), 1046–1070. https://doi.org/10.1080/14616696.2023.2277281
- Schroeder, D. A., Graziano, W. G., & Davis, M. H. (2015). Empathy and Prosocial Behavior. In D. A. Schroeder & W. G. Graziano (Eds.), *The Oxford Handbook of Prosocial Behavior*. Oxford University Press. https://doi.org/10.1093/oxfordhb/9780195399813.013.026
- Shupp, R., Loveridge, S., Skidmore, M., Lim, J., & Rogers, C. (2017). Trust and Patience after a Tornado. *Weather, Climate, and Society*, 9(4), 659–668. https://doi.org/10.1175/WCAS-D-16-0135.1
- Sisco, M. R. (2021). The effects of weather experiences on climate change attitudes and behaviors. *Current Opinion in Environmental Sustainability*, *52*, 111–117. https://doi.org/10.1016/j.cosust.2021.09.001
- Solnit, R. (2009). A paradise built in hell: The extraordinary communities that arise in disasters. Viking.
- Spektor, M., Fasolin, G. N., & Camargo, J. (2023). Climate change beliefs and their correlates in Latin America. *Nature Communications*, 14(1), 7241. https://doi.org/10.1038/s41467-023-42729-x
- Spence, A., Poortinga, W., Butler, C., & Pidgeon, N. F. (2011). Perceptions of climate change and willingness to save energy related to flood experience. *Nature Climate Change*, 1(1), 46–49. https://doi.org/10.1038/nclimate1059
- Statistisches Bundesamt [Destatis]. (2023). Auszug aus dem Gemeindeverzeichnis Gebietsstand: 31.12.2022 [Data set] [Dataset]. https://www.destatis.de/DE/Themen/Laender-

Regionales/Gemeindeverzeichnis/Administrativ/Archiv/GVAuszugJ/31122022_Auszu g_GV.html

- Steimanis, I., & Vollan, B. (2022). Prosociality as response to slow- and fast-onset climate hazards. *Global Sustainability*, *5*, e10-undefined. https://doi.org/10.1017/sus.2022.9
- Swim, J., Clayton, S., Doherty, T., Gifford, R., Howard, G., Reser, J., Stern, P., & Weber, E. (2009). Psychology and global climate change: Addressing a multi-faceted phenomenon and set of challenges. A report by the American Psychological Association's task force on the interface between psychology and global climate change. *American Psychological Association, Washington*, 66, 241–250.
- Tradowsky, J. S., Philip, S. Y., Kreienkamp, F., Kew, S. F., Lorenz, P., Arrighi, J., Bettmann, T., Caluwaerts, S., Chan, S. C., De Cruz, L., De Vries, H., Demuth, N., Ferrone, A., Fischer, E. M., Fowler, H. J., Goergen, K., Heinrich, D., Henrichs, Y., Kaspar, F., ... Wanders, N. (2023). Attribution of the heavy rainfall events leading to severe flooding in Western Europe during July 2021. *Climatic Change*, 176(7), 90. https://doi.org/10.1007/s10584-023-03502-7
- Van Der Linden, S. (2014). On the relationship between personal experience, affect and risk perception: The case of climate change. *European Journal of Social Psychology*, 44(5), 430–440. https://doi.org/10.1002/ejsp.2008
- Vardy, T., & Atkinson, Q. D. (2019). Property Damage and Exposure to Other People in Distress Differentially Predict Prosocial Behavior After a Natural Disaster. *Psychological Science*, 30(4), 563–575. https://doi.org/10.1177/0956797619826972
- Weber, E. U. (2010). What shapes perceptions of climate change? *WIREs Climate Change*, *1*(3), 332–342. https://doi.org/10.1002/wcc.41
- Whitmarsh, L. (2008). Are flood victims more concerned about climate change than other people? The role of direct experience in risk perception and behavioural response. *Journal of Risk Research*, 11(3), 351–374. https://doi.org/10.1080/13669870701552235
- Yamamura, E. (2016). Natural disasters and social capital formation: The impact of the Great Hanshin-Awaji earthquake: Natural disasters and social capital. *Papers in Regional Science*, 95, S143–S164. https://doi.org/10.1111/pirs.12121

Supplementary Information for

Natural Disaster Experience Does Not Affect Environmental Attitudes or Prosociality

Evidence from the 2021 Flood in Germany

Authors: Nils Christian Hoenow^{1,*}, Kiran Karki¹, Maximilian Nicolaus Burger²

Affiliations: ¹RWI – Leibniz Institute for Economic Research, Hohenzollernstr. 1-3 45128 Essen, Germany, ²Department of Economics, Philipps University Marburg, 35032 Marburg, Germany ***Corresponding author:** christian.hoenow@rwi-essen.de

Contents of Supplementary Information:

Tables:

A. Respondents' socioeconomic characteristics across data sets and flood exposure

- B. Power analysis
 - 1) Environmental attitudes
 - 2) Environmental behavior
 - 3) Support for climate policies
 - 4) Prosociality
- C. Regressions with covariates
 - 1) Environmental attitudes
 - 2) Environmental behavior
 - 3) Support for climate policies
 - 4) Prosociality

D. Differences-in-Differences Analysis

- 1) Panel variables in data set 1
- 2) Panel variables in data set 2

E. Severe exposure

- 1) Environmental attitudes
- 2) Environmental behavior
- 3) Support for climate policies
- 4) Prosociality
- F. Self-reported exposure (and degrees of affectedness)
 - 1) Environmental attitudes, behavior and support for climate policies
 - 2) Tests for differences (environment variables)
 - 3) Tests for differences (procosociality variables)

Supplements

- Q. Survey questions
 - 1) Data set 1 (RWI Climate-Mobility Panel)
 - 2) Data set 2 (Socio-Ecological Panel)
 - 3) Data set 3 (NEWCOMERS)

I. Information on availability of analysis code and data

	data (Climate-Mo	set 1 bility Panel)	data (Socio-Ecolo	set 2 ogical Panel)	data (NEWC)	set 3 OMERS)	German general population
	Not exposed	Exposed	Not exposed	Exposed	Not exposed	Exposed	taken from Mikrozensus
	n = 5,135 (81.70 %)	n = 1,150 (18.30 %)	n = 4,843 (80.68 %)	n = 1,160 (19.32 %)	n = 1,179 (80.42 %)	n = 287 (19.58 %)	values of December 2021
Gender							C: 12411-0005
Female	46.10 %	47.13 %	43.80 %	41.21 %	49.02 %	49.13 %	50.66 %
Male	53.90 %	52.87 %	56.08 %	58.53 %	50.98 %	50.87 %	49.34 %
Non-binary	0.00 %	0.00 %	0.12 %	0.26 %	0.00 %	0.00 %	0.00 %
Age Category							C: 12411-0005
18-29 Years	5.96 %	5.22 %	3.49 %	2.93 %	18.32 %	11.15 %	15.84 %
30-39 Years	12.13 %	10.70 %	10.57 %	8.97 %	19.25 %	17.42 %	15.70 %
40-49 Years	15.38 %	14.52 %	14.37 %	12.41 %	17.98 %	18.12 %	14.41 %
50-59 Years	22.98 %	22.96 %	22.28 %	23.36 %	24.94 %	33.80 %	18.84 %
60+ Years	43.54 %	46.61 %	49.29 %	52.33 %	19.51 %	19.51 %	35.22 %
Income Category							C: 12211-0300
No response	11.35 %	10.87 %	10.26 %	10.69 %	0.00 %	0.00 %	0.63%
Low	4.48 %	3.13 %	5.18 %	5.17 %	13.49 %	13.94 %	15.66%
Medium	26.66 %	31.30 %	26.31 %	28.53 %	36.05 %	36.59 %	32.89%
High	29.19 %	29.30 %	29.96 %	28.10 %	36.56 %	34.84 %	26.03%
Very high	28.32 %	25.39 %	28.29 %	27.50 %	13.91 %	14.63 %	24.79%
High Education							C: 12211-0101
No response	2.63 %	2.61 %	2.31 %	2.16 %			0.16%
Yes	27.09 %	23.57 %	38.70 %	35.17 %	24.00 %	17.42 %	19.48%
No	70.28 %	73.83 %	58.99 %	62.67 %	76.00 %	82.58 %	80.36%
Household Size							C: 12211-0300
No response	0.47 %	0.26 %	0.47 %	0.43 %	0.00 %	0.00 %	0.00 %
1	25.67 %	25.91 %	26.72 %	27.50 %	22.82 %	21.60 %	41.68%
2	46.56 %	46.26 %	49.33 %	50.09 %	44.78 %	44.25 %	33.47%
3	13.55 %	12.43 %	11.67 %	11.55 %	17.39 %	20.56 %	11.83%
4+	13.75 %	15.13 %	11.81 %	10.43 %	15.01 %	13.59 %	13.02%
East Germany							C: 12211-1001
Yes	21.17 %	7.30 %	23.83 %	14.31 %	23.41 %	8.71 %	19.36 %
No	78 83 %	92 70 %	76 17 %	85 69 %	76 59 %	01 20 %	80.64.%

A.1 Respondents' socioeconomic characteristics aacross data sets and flood exposure

Note: Maximum age in data set 3 was 69, consequently, the last age category is smaller compared to other data sets and thus, the shares of the other age categories are bigger, monthly net household income: low income = less than 1,200 EUR, medium income = less than 2,700 EUR, high income = less than 4,200 EUR, very high income = more than 4,200 EUR, the income categories for data set 3 are slightly higher (50 EUR for each category), monthly net household income for the German population: low income = less than 1,250 EUR, medium income = less than 2,500 EUR, high income = less than 4,000 EUR, very high income = more than 4,000 EUR. C = Code for variable in micro census.

B.1 Power analysis for environmental attitudes

Dependent Variables	Response options	Given sample size	Standard deviation / proportion in given sample	MDE for given std. dev.
data set 1		5,135 / 1,150		
Impact of climate change	Five-point scale	4,966 / 1,110	0.704	0.058
data set 2		4,842 / 1,160		
Existence of climate change ^C	Yes/No	4,781 / 1,146	0.927	0.021
Climate change manmade ^C	Yes/No	4,764 / 1,141	0.881	0.027
Dangers of climate change	Four-point scale	4,427 / 1,056	0.667	0.057
Importance of climate policy	Four-point scale	4,823 / 1,153	0.735	0.060
Relative importance ^C	Yes/No	4,843 / 1,160	0.444	0.040
data set 3		1,179 / 287		
Seriousness of climate change	Ten-point scale	1,179 / 287	2.410	0.393
Importance of protecting the environment	Four-point scale	1,179 / 287	0.663	0.108

MDEs are calculated for significance levels of 10% (alpha = 0.1) and 80% power (1-beta = 0.80). C = variable is dichotomous

B.2 Power analysis for environmental behavior

Dependent Variables	Response options	Given sample size	Standard deviation / proportion in given sample	MDE for given std. dev.
data set 1		5,135 / 1,150		
Environmentally friendly behavior	Five-point scale	5,082 / 1,140	0.779	0.063
Green Party preference ^C	Yes/No	4,819 / 1,080	0.184	0.032
data set 2		4,842 / 1,160		
Green Party preference ^C	Yes/No	4,639 / 1,123	0.234	0.035
data set 3		1,179 / 287		
Conserving energy at home	Five-point scale	1,179 / 287	0.840	0.137
Donation to atmosfair	Value between 0 and 100	1,179 / 287	30.837	5.047

MDEs are calculated for significance levels of 10% ($\alpha = 0.10$) and 80% power (1- $\beta = 0.80$). C = variable is dichotomous

B.3 Power analysis for support for climate policies

Dependent Variables	Response options	Given sample size	Standard deviation / proportion in given sample	MDE for given std. dev.
data set 1		5,135 / 1,150		
Increase in carbon pricing	Five-point scale	3,958 / 923	1.313	0.119
No new vehicles with combustion engines after 2035	Five-point scale	4,990 / 1,129	1.382	0.113
data set 2		4,842 / 1,160		
No new vehicles with combustion engines after 2030	Five-point scale	4,755 / 1,144	1.462	0.120
Constant increase in tax on petrol and diesel	Five-point scale	4,715 / 1,132	1.284	0.106
Speed limit of 130km/h on highways	Five-point scale	4,817 / 1,154	1.551	0.126
data set 3		1,179 / 287		
Coal phase-out	Ten-point scale	1,179 / 287	1.155	0.189
Expansion of electric vehicles	Five-point scale	1,179 / 287	1.283	0.210
Expansion of public transport	Five-point scale	1,179 / 287	0.927	0.152

MDEs are calculated for significance levels of 10% ($\alpha = 0.10$) and 80% power (1- $\beta = 0.80$). C = variable is dichotomous

B.4 Power analysis for prosociality

Dependent Variables	Response options	Given sample size	Standard deviation / proportion in given sample	MDE for given std. dev.
data set 1		5,135 / 1,150		
Positive reciprocity	Eleven-point scale	5,027 / 1,128	1.721	0.141
Negative reciprocity	Eleven-point scale	4,982 / 1,117	2.444	0.201
Trust	Eleven-point scale	4,967 / 1,118	2.338	0.192
Altruism	Eleven-point scale	5,008 / 1,123	2.491	0.204
Dictator game donations	Tokens 0 - 10	3,669 / 869	1.472	0.138
Past donation ^C	Yes/No	4,961 / 1,129	0.805	0.033
Past donations (amount)	Values ≥ 0	4,411 / 980	540.339	47.446
Volunteer work ^C	Yes/No	5,044 / 1,130	0.426	0.040

MDEs are calculated for significance levels of 10% (α = 0.10) and 80% power (1- β = 0.80). C = variable is dichotomous

C.1 Regression analyses for environmental attitudes

	Dependent variables								
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
	Impact of	Existence of	Climate	Dangers of	Importance of	Relative	Seriousness of	Importance of	
	climate	climate	change	climate	climate policy i	mportance (d	l)climate change	protecting the	
	cnange	change (d)	(d)	cnange				environment	
data set	1	2	2	2	2	2	3	3	
Exposed to the flood (d)	-0.024	-0.006	-0.018	-0.022	-0.054**	-0.026	-0.281*	-0.070	
	(0.341)	(0.525)	(0.123)	(0.361)	(0.037)	(0.131)	(0.085)	(0.100)	
Female (d)	0.095***	-0.028***	0.010	0.103***	0.165***	0.037***	0.352***	0.006	
	(<0.001)	(0.263)	(<0.001)	(<0.001)	(<0.001)	(0.007)	(0.008)	(0.859)	
Age category									
30-39 years	-0.136***	-0.039**	-0.037	-0.094*	-0.067	-0.122***	-0.027	0.053	
	(0.006)	(0.031)	(0.110)	(0.055)	(0.257)	(0.004)	(0.890)	(0.364)	
40-49 years	-0.219***	-0.015	-0.030	-0.162***	-0.062	-0.195***	-0.149	0.048	
	(<0.001)	(0.406)	(0.198)	(0.001)	(0.307)	(<0.001)	(0.485)	(0.425)	
50-59 years	-0.205***	-0.038**	-0.056***	-0.193***	-0.069	-0.179***	0.069	0.089	
	(<0.001)	(0.022)	(0.009)	(<0.001)	(0.217)	(<0.001)	(0.724)	(0.118)	
60+ years	-0.315***	-0.018	-0.032	-0.160***	0.011	-0.145***	0.040	0.078	
	(<0.001)	(0.233)	(0.106)	(<0.001)	(0.841)	(<0.001)	(0.854)	(0.205)	
Income category									
Medium	-0.032	0.040**	0.049**	-0.038	0.058	0.049	-0.353*	-0.102*	
	(0.564)	(0.044)	(0.036)	(0.392)	(0.274)	(0.107)	(0.088)	(0.097)	
High	-0.037	0.030	0.045*	0.001	0.094*	0.046	-0.212	-0.08	
	(0.524)	(0.155)	(0.066)	(0.976)	(0.089)	(0.150)	(0.329)	(0.200)	
Very high	-0.015	0.061***	0.079***	0.027	0.154***	0.073**	-0.181	-0.124	
	(0.804)	(0.004)	(0.002)	(0.582)	(0.007)	(0.029)	(0.495)	(0.106)	
High education (d)	0.089***	0.052***	0.061***	0.066***	0.164***	0.125***	0.439***	0.146***	
	(<0.001)	(<0.001)	(<0.001)	(0.001)	(<0.001)	(<0.001)	(0.005)	(0.001)	
Household size									
2	0.001	-0.016*	-0.031***	-0.048*	-0.095***	-0.059***	0.032	0.058	
	(0.964)	(0.092)	(0.007)	(0.052)	(<0.001)	(0.001)	(0.861)	(0.243)	
3	-0.012	-0.030**	-0.056***	-0.101***	-0.116***	-0.078***	0.040	0.079	
	(0.739)	(0.028)	(0.001)	(0.010)	(0.004)	(0.003)	(0.854)	(0.196)	
4+	0.042	-0.031**	-0.054***	-0.091**	-0.164***	-0.048*	0.293	0.141**	
	(0.261)	(0.031)	(0.003)	(0.021)	(<0.001)	(0.089)	(0.197)	(0.020)	
East Germany (d)	-0.070***	< 0.001	-0.024**	-0.120***	-0.203***	-0.104***	-0.330**	-0.012	
• • •	(0.005)	(0.974)	(0.032)	(<0.001)	(<0.001)	(<0.001)	(0.039)	(0.779)	
Constant	4.006***	0.922***	0.878***	3.782***	3.602***	0.560***	7.434***	3.189***	
	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	
R-squared	0.028	0.022	0.019	0.021	0.046	0.034	0.019	0.015	
Observations	5.309	5.217	5.198	4,850	5.257	5,272	1,466	1,466	

	Dependent variables							
-	(1)	(2)	(3)	(4)	(5)			
	Environmentally	Green Party	Green Party	Conserving energy	Donation			
	friendly behavior	Preference (d)	Preference (d)	at home	to atmosfair			
data set	1	1	2	3	3			
Exposed to the flood (d)	-0.012	-0.012	-0.022	0.024	-1.499			
	(0.657)	(0.361)	(0.125)	(0.664)	(0.476)			
Female (d)	0.101***	0.091***	0.082***	0.054	2.724			
	(<0.001)	(<0.001)	(<0.001)	(0.237)	(0.108)			
Age category								
30-39 years	-0.056	-0.062**	-0.076*	0.154**	-1.757			
	(0.339)	(0.041)	(0.072)	(0.044)	(0.492)			
40-49 years	-0.100*	-0.072**	-0.136***	0.261***	0.324			
	(0.078)	(0.016)	(0.001)	(<0.001)	(0.903)			
50-59 years	-0.055	-0.065**	-0.158***	0.298***	2.135			
	(0.313)	(0.023)	(<0.001)	(<0.001)	(0.400)			
60+ years	0.097*	-0.107***	-0.186***	0.271***	5.786**			
	(0.061)	(<0.001)	(<0.001)	(0.001)	(0.041)			
Income category								
Medium	0.007	-0.028	0.002	0.078	-0.627			
	(0.914)	(0.299)	(0.945)	(0.320)	(0.814)			
High	0.063	-0.013	0.005	0.145*	4.828*			
	(0.340)	(0.653)	(0.870)	(0.076)	(0.088)			
Very high	0.102	0.002	0.012	0.124	6.870**			
	(0.142)	(0.954)	(0.683)	(0.210)	(0.040)			
High education (d)	0.161***	0.149***	0.113***	0.140***	6.662***			
	(<0.001)	(<0.001)	(<0.001)	(0.010)	(0.001)			
Household size								
2	-0.002	-0.013	-0.017	-0.119*	-2.256			
	(0.947)	(0.383)	(0.269)	(0.059)	(0.337)			
3	-0.022	-0.022	-0.028	-0.172**	-3.423			
	(0.601)	(0.259)	(0.215)	(0.028)	(0.221)			
4+	0.038	0.007	-0.012	-0.041	-6.535**			
	(0.389)	(0.726)	(0.616)	(0.600)	(0.021)			
East Germany (d)	-0.108***	-0.079***	-0.108***	-0.043	-2.253			
	(<0.001)	(<0.001)	(<0.001)	(0.460)	(0.272)			
Constant	3.205***	0.226***	0.350***	3.739***	34.990***			
	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)			
R-squared	0.026	0.058	0.051	0.027	0.029			
Observations	5,426	5,245	5,119	1,466	1,466			

C.2 Regression analyses for environmental behavior

				Depende	nt variables			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	Increase in	Ban on	Ban on	Constant	Speed	Coal	Expansion	Expansion
	carbon	combustion	combustion	increase in	limit of	phase-out	of electric	of public
data set	pricing 1	1 car 2055	2050 cars 2050	2	130km/n	3	venicies	transport
Exposed to the flood (d)	0.055	0.061	0.055	0.053	0.056	0.130*	0.021	0.118*
Exposed to the flood (d)	-0.055	-0.001	-0.055	-0.033	-0.030	-0.130	-0.021	-0.118
Fomala (d)	(0.270)	(0.201)	(0.207)	(0.231)	-0.293	(0.080)	(0.810)	(0.004)
Telliale (u)	(<0.001)	(<0.001)	(<0.001)	(0.453)	(< 0.001)	-0.133	-0.104	-0.011
A go ostogomi	(<0.001)	(<0.001)	(<0.001)	(0.455)	(<0.001)	(0.037)	(0.140)	(0.855)
20 20 voors	0 240***	0.240**	0.228*	0 229**	0.052	0 161	0.022	0.044
50-59 years	-0.340***	-0.249	-0.238	-0.338**	0.032	-0.101	-0.022	-0.044
40.40 voors	(0.001)	0.407***	(0.080)	(0.010)	-0.717	(0.104)	(0.642)	(0.399)
40-49 years	(<0.001)	-0.40700	-0.48700	-0.41/***	0.080	-0.243	-0.302***	-0.032
50, 50 years	(<0.001)	(<0.001)	(<0.001)	0.540***	-0.343	(0.018)	0.371***	0.046
50-59 years	(< 0.001)	-0.442	(< 0.001)	(< 0.001)	0.191	-0.230^{11}	-0.371***	(0.565)
60 L Maars	0.286***	(<0.001)	(<0.001)	(<0.001)	-0.149	0.307***	0.430***	(0.505)
00+ years	(0.002)	(0.019)	(0.002)	(<0.001)	(< 0.001)	(0.004)	(< 0.001)	(0.599)
Income category	(0.002)	(0.01))	(0.002)	(<0.001)	(<0.001)	(0.004)	(<0.001)	(0.377)
Medium	-0 195*	-0.215**	-0 203**	-0 183**	-0.129	-0 179*	-0.105	-0.115
Weddulli	(0.068)	(0.031)	(0.042)	(0.043)	-0.129	(0.074)	(0.331)	(0.137)
High	-0.151	-0 232**	-0 272***	-0 114	-0 271***	-0.098	-0.062	-0.033
Ingh	(0.176)	(0.026)	(0,009)	(0.224)	-0.008	(0.346)	(0.587)	(0.684)
Very high	-0.045	-0.235**	-0.186*	-0.082	-0.373***	-0.083	0.049	0.048
, or y mgn	(0.699)	(0.031)	(0.086)	(0.406)	-0.001	(0.517)	(0.726)	(0.626)
High education (d)	0.563***	0.449***	0.432***	0.394***	0.356***	0.335***	0.342***	0.106*
8(-)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(0.083)
Household size	()	(((((((00000)
2	-0.169***	-0.149***	-0.169***	-0.164***	0.118**	-0.060	-0.066	-0.128*
	(0.002)	(0.004)	(0.002)	(0.001)	-0.035	(0.490)	(0.486)	(0.066)
3	-0.216***	-0.216***	-0.184**	-0.164**	0.076	-0.114	-0.204*	-0.229***
	(0.004)	(0.003)	(0.018)	(0.018)	-0.375	(0.271)	(0.076)	(0.005)
4+	-0.046	-0.162**	-0.184**	-0.104	0.155*	-0.164	-0.228*	-0.286***
	(0.564)	(0.034)	(0.028)	(0.169)	-0.087	(0.138)	(0.056)	(0.001)
East Germany (d)	-0.277***	-0.182***	-0.203***	-0.159***	-0.213***	-0.295***	-0.021	-0.010
• • •	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(0.804)	(0.881)
Constant	3.249***	3.144***	3.183***	2.755***	3.340***	4.127***	3.584***	4.249***
	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)
R-squared	0.063	0.039	0.041	0.038	0.054	0.038	0.039	0.022
Observations	4,268	5,337	5,197	5,150	5,255	1,466	1,466	1,466

C.3 Regression analyses for support for climate policies

				Depende	ent variables			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	Positive	Negative	Trust	Altruism	Dictator game	Past donation	s Past donations	Volunteer
	reciprocity	reciprocity			donation	(d)	(amount)	work (d)
data set	1	1	1	1	1	1	1	1
Exposed to the flood (d)	0.087	0.020	-0.015	0.125	0.049	0.006	15.545	-0.013
	(0.125)	(0.816)	(0.853)	(0.140)	(0.425)	(0.636)	(0.416)	(0.464)
Female (d)	0.011	-0.776***	0.436***	0.715***	0.063	0.067***	12.849	-0.051***
	(0.817)	(<0.001)	(<0.001)	(<0.001)	(0.172)	(<0.001)	(0.381)	(<0.001)
Age category								
30-39 years	-0.036	0.028	-0.088	-0.147	0.031	-0.038	25.310	-0.044
	(0.725)	(0.860)	(0.608)	(0.384)	(0.752)	(0.221)	(0.336)	(0.196)
40-49 years	-0.028	0.042	0.101	0.220	0.164*	0.036	129.277***	0.007
	(0.786)	(0.786)	(0.543)	(0.178)	(0.094)	(0.234)	(<0.001)	(0.822)
50-59 years	0.035	-0.124	0.399**	0.365**	0.287***	0.062**	148.136***	0.035
	(0.714)	(0.406)	(0.013)	(0.021)	(0.002)	(0.028)	(<0.001)	(0.267)
60+ years	-0.124	-0.049	0.548***	1.044***	0.442***	0.143***	295.436***	0.128***
	(0.175)	(0.727)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)
Income category								
Medium	-0.019	-0.078	0.160	0.655***	0.214*	0.178***	124.789***	0.019
	(0.881)	(0.662)	(0.352)	(<0.001)	(0.089)	(<0.001)	(<0.001)	(0.552)
High	-0.031	-0.144	0.189	0.977***	0.236*	0.244***	245.835***	0.046
	(0.818)	(0.443)	(0.289)	(<0.001)	(0.079)	(<0.001)	(<0.001)	(0.183)
Very high	-0.039	0.091	0.155	1.382***	0.309**	0.274***	364.361***	0.065*
	(0.783)	(0.644)	(0.404)	(<0.001)	(0.028)	(<0.001)	(<0.001)	(0.070)
High education (d)	0.009	-0.164**	0.185**	0.368***	-0.115**	0.083***	158.175***	0.089***
	(0.869)	(0.036)	(0.015)	(<0.001)	(0.036)	(<0.001)	(<0.001)	(<0.001)
Household size								
2	0.028	0.037	0.126	-0.258***	-0.005	-0.023	-59.430***	0.011
	(0.668)	(0.690)	(0.154)	(0.005)	(0.941)	(0.112)	(0.006)	(0.564)
3	0.103	0.189	0.159	-0.451***	-0.055	-0.018	-119.165***	0.061**
	(0.217)	(0.137)	(0.186)	(<0.001)	(0.550)	(0.370)	(<0.001)	(0.015)
4+	0.013	-0.080	0.335***	-0.339***	0.074	-0.003	-75.063**	0.117***
	(0.888)	(0.536)	(0.009)	(0.010)	(0.427)	(0.884)	(0.013)	(<0.001)
East Germany (d)	0.041	0.037	-0.091	-0.318***	-0.006	-0.074***	-97.359***	-0.075***
• • •	(0.486)	(0.672)	(0.280)	(<0.001)	(0.927)	(<0.001)	(<0.001)	(<0.001)
Constant	9.418***	4.002***	5.423***	6.289***	4.635***	0.484***	-85.494***	0.314***
	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(<0.001)	(0.002)	(<0.001)
R-squared	0.002	0.029	0.019	0.073	0.016	0.063	0.088	0.029
Observations	5,396	5,366	5,355	5,381	4,025	5,335	4,851	5,412

C.4 Regression analyses for prosociality

D.1 Differences-in-differences for data set 1 (RWI Climate-Mobility Panel): June 2019 and March/April 2022

data set 1	2019		20	2022		Time Trend	
	Not exposed	Exposed	Not exposed	Exposed	Not exposed	Exposed	Estimate
Perceived impact of climate change	3.831 (0.678)	3.780 (0.693)	3.785 (0.674)	3.753 (0.678)	-0.046*** (0.004)	-0.027 (0.441)	0.018 (0.620)
Green Party	[2,293] 0.163	[514] 0.150	[2,293] 0.173	[514] 0.156	[2,293] 0.010* (0.087)	[514] 0.006	-0.005
preference (d)	[2,304]	[525]	[2,304]	[525]	[2,304]	[525]	(0.735)
Donation (d)	0.761 - [2,320]	0.762 [534]	0.812	0.830 - [534]	0.050*** (<0.001) [2,320]	0.067*** (<0.001) [534]	0.017 (0.361)

Standard deviation in round brackets for means, p-values in round brackets for time trend (within-subject differences) and DiD. Number of observations in square brackets. Time trend tests for significance by paired t-tests or McNemar if variable is dichotomous. (d) denotes dichotomous variables. Significance levels: p < 0.10, p < 0.05, p < 0.01. DiD-estimate based on Fixed-effects estimation: the respective variable is regressed on the survey wave (dummy for 2022), on the treatment (dummy for being affected), as well as on the interaction of wave and treatment. The results equal a t-test across the time trends for each group.

D.2 Differences-in-differences for data set 2 (Socio-Ecological Panel): June 2021 and July/August

2022

data set 2	2021		20	2022 Tin		Trend	DiD
	Not exposed	Exposed	Not exposed	Exposed	Not exposed	Exposed	Estimate
Existence of climate change (d)	0.900 - [4,703]	0.904 - [1,129]	0.930 - [4,703]	0.923	0.030*** (<0.001) [4,703]	0.019* (0.058) [1,129]	-0.012 (0.282)
Climate change manmade (d)	0.860 - [4,670]	0.849 - [1,119]	0.885 - [4,760]	0.867 - [1,119]	0.025*** (<0.001) [4,670]	0.018* (0.079) [1,119]	-0.007 (0.546)
Relative importance (d)	0.574 [4,823]	0.566 - [1,153]	0.446 [4,823]	0.435 [1,153]	-0.128*** (<0.001) [4,823]	-0.131*** (<0.001) [1,153]	-0.003 (0.865)
Green Party preference (d)	0.212 [4,527]	0.191 - [1,092]	0.240 [4,527]	0.219 [1,092]	0.028*** (<0.001) [4,527]	0.027*** (<0.001) [1,092]	<0.001 (0.989)

Standard deviation in round brackets for means, p-values in round brackets for time trend (within-subject differences) and DiD. Number of observations in square brackets. Time trend tests for significance by paired t-tests or McNemar if variable is dichotomous. (d) denotes dichotomous variables. Significance levels: p < 0.10, p < 0.05, p < 0.01. DiD-estimate based on Fixed-effects estimation: the respective variable is regressed on the survey wave (dummy for 2022), on the treatment (dummy for being affected), as well as on the interaction of wave and treatment. The results equal a t-test across the time trends for each group.

E.1 Severe exposure and environmental attitudes

			Not exposed	Severely exposed	
Survey	Question/Statement	Response option	means (std. dev.) [n]		difference (p-value)
data set 1					
Impact of climate change	What do you think the consequences of climate change will be for your personal living conditions in the coming decades?	Five-point scale (very negative - very positive consequences)	3.791 (0.702) [5,881]	3.790 (0.754) [195]	-0.001 (0.977)
data set 2					
Existence of climate change	Based on everything you know; do you think the world's climate is changing or not?	Yes/No	0.928	0.914 - [185]	-0.014 (0.458) ^C
Climate change manmade	Do you think climate change is mostly caused by humans (as opposed to mostly caused by natural factors)?	Yes/No	0.881	0.8750	-0.006 (0.800) ^C
Dangers of climate change	Do you think climate change is dangerous for humanity?	Four-point scale (certainly not - certainly)	3.605 (0.669) [5,314]	3.663 (0.596) [169]	-0.058 (0.268)
Importance of climate policy	Do you consider climate policy to be an important field in politics?	Four-point scale (not important - important)	3.660 (0.734) [5,789]	3.652 (0.756) [1,153]	-0.008 (0.889)
Relative importance	In your opinion, are the environment and climate change one of the most important issues Germany is currently facing?	Yes/No	0.444 	0.436 - [188]	-0.008 (0.831) ^C
data set 3					
Seriousness of climate change	How serious a problem do you think climate change is at this moment?	Ten-point scale (not at all a serious problem - an extremely serious problem)	7.414 (2.410) [1,412]	7.370 (2.436) [54]	-0.044 (0.895)
Importance of protecting the environment	How important is protecting the environment to you personally?	Four-point scale (not at all important - very important)	3.250 (0.665) [1,412]	3.111 (0.572) [54]	-0.139 (0.131)
For the calculation of means a Γ wo-sided t-tests for metric ar $C = Chi$ -squared test if variabl Significance levels: * $p < 0.10$	nd proportions, "Do not know" and ad response-scale variables e is dichotomous b, ** $p < 0.05$, *** $p < 0.01$	"Prefer not to say" responses	were omitted.		

E.2 Severe exposure and environmental behavior

			Not exposed	Severely exposed	
Survey	Question/Statement	Response options	means (std. dev.) [n]		difference (p-value)
data set 1					
Environmentally friendly behavior	I behave in an environmentally conscious manner, even if this is associated with higher costs and/or efforts.	Five-point scale (strongly disagree - strongly agree)	3.329 (0.780) [6,022]	3.410 (0.731) [200]	-0.081 (0.149)
Green Party preference	If you lean towards a certain party, do you have a preference for the Green Party?	Yes/No	0.184 - [5,704]	0.179 - [195]	-0.005 (0.861) ^C
data set 2					
Green Party preference	If you lean towards a certain party, do you have a preference for the Green Party?	Yes/No	0.236 [5,580]	0.192 [182]	-0.043 (0.174) ^C
data set 3					
Conserving energy at home	How much do you agree or disagree with the following statements? "I take action to conserve energy at home"	Five-point scale (strongly disagree - strongly agree	4.008 (0.841) [1,412]	4.056 (0.811) [54]	-0.048 (0.682)
Donation to atmosfair	Donation Experiment: How much of 100 € would you like to donate to atmosfair?	Values between 0 and 100	38.620 (30.756) [1,412]	32.852 (32.686) [54]	-5.769 (0.177)

For the calculation of means and proportions, "Do not know" and "Prefer not to say" responses were omitted. Two-sided t-tests for metric and response-scale variables C = Chi-squared test if variable is dichotomous Significance levels: * p < 0.10, ** p < 0.05, *** p < 0.01

E.3 Severe exposure and support of climate policies

			Not exposed	Severely exposed	
Survey	Question/Statement	- Response options	means (std. dev.)		difference (p-value)
data set 1	Question power interior]	
Increase in carbon pricing	What is your stance on the following measure?	Five-point scale (strongly disagree – strongly agree)	2.837 (1.311) [4,714]	2.886 (1.364) [167]	-0.050 (0.632)
No new vehicles with combustion engines after 2035	What is your stance on the following measure?	Five-point scale (strongly disagree – strongly agree)	2.620 (1.382) [5,920]	2.653 (1.373) [199]	-0.034 (0.737)
data set 2					
No new vehicles with combustion engines after 2030 ^A	Do you support the following measure?	Five-point scale (No - Yes)	2.592 (1.460) [5,713]	2.683 (1.507) [186]	-0.091 (0.418)
Constant increase in tax on petrol and diesel ^A	Do you support the following measure?	Five-point scale (No - Yes)	2.175 (1.283) [5,662]	2.216 (1.322) [185]	-0.041 (0.677)
Speed limit of 130km/h on highways ^A	Do you support the following measure?	Five-point scale (No - Yes)	3.771 (1.549) [5,783]	3.814 (1.610) [188]	-0.043 (0.718)
data set 3					
Coal phase-out	Please indicate what your opinion is on the following policy measure.	Ten-point scale (not at all a serious problem - an extremely serious problem)	3.682 (1.157) [1,412]	3.389 (1.071) [54]	-0.293* (0.067)
Expansion of electric vehicles	Please indicate what your opinion is on the following policy measure.	Five-point scale (strongly disagree – strongly agree)	3.212 (1.281) [1,412]	3.074 (1.344) [54]	-0.138 (0.439)
Expansion of public transport	Please indicate what your opinion is on the following policy measure	Five-point scale (strongly disagree – strongly agree)	4.067 (0.925) [1.412]	3.944 (0.979) [54]	-0.12 (0.339)

A = The question was preceded by an introductory statement: "For the mobility transition there are some measures being discussed that aim to reduce car use and thereby emission of pollutants and carbon dioxide." Significance levels: p < 0.10, ** p < 0.05, *** p < 0.01

E.4 Severe exposure and prosociality

			Not exposed	Severely exposed	
Survey	Question/Statement	Response options	means (std. dev.) [n]		difference (p-value)
data set 1					
Positive reciprocity	Willingness to return a favor	Eleven-point scale	9.371 (1.722) [5,957]	9.303 (1.701) [198]	-0.068 (0.582)
Negative reciprocity	Willingness to take revenge	Eleven-point scale	3.576 (2.448) [5,902]	3.315 (2.293) [197]	-0.262 (0.139)
Trust	"People have only the best intentions"	Eleven-point scale	6.257 (2.344) [5,888]	6.137 (2.166) [1,118]	-0.120 (0.477)
Altruism	Willingness to give to good causes	Eleven-point scale	7.937 (2.491) [5,933]	8.202 (2.491) [198]	-0.265 (0.141)
Dictator game donation	Indicate how many tokens you would like to give to your unknown partner	Tokens between 0 and 10	5.165 (1.474) [4,382]	5.218 (1.420) [156]	-0.053 (0.657)
Past donations	Donation last year	Yes/No	0.804 - [5,890]	0.820 - [200]	-0.016 (0.577) ^C
Past donations (amount)	Donation amount	Values ≥ 0	313.747 (543.703) [5,218]	328.630 (427.616) [173]	-14.883 (0.722)
Volunteer work	Volunteering	Yes/No	0.424 - [5,975]	0.477 - [199]	-0.053 (0.136) ^C

For the calculation of means and proportions, "Do not know" and "Prefer not to say" responses were omitted. Two-sided t-tests for metric and response-scale variables C = Chi-squared test if variable is dichotomous Significance levels: * p < 0.10, ** p < 0.05, *** p < 0.01

F.1 Self-reported degrees of affectedness on environmental variables

	(1)	(2)	(3)	(4)
Affectedness	Not affected		Affected	
		All	W/o damage	W/ damage
Self-reported exposure:	no	yes	yes	yes
Damage:			no	yes
ENV variable:				
Impact of Climate Change	3.791	3.824	3.860	3.746
	(0.702)	(0.740)	(0.698)	(0.823)
Env. friendly behavior	3.331	3.390	3.446	3.271
	(0.778)	(0.820)	(0.740)	(0.962)
Green Party preference (d)	0.185	0.180	0.171	0.197
	-	-	-	-
Increase in carbon pricing	2.838	2.881	2.931	2.783
	(1.314)	(1.306)	(1.214)	(1.474)
No new vehicles with combustion	2.622	2.634	2.667	2.565
engines after 2035	(1.384)	(1.323)	(1.305)	(1.366)
Observations	6,043	218	148	70

Means shown for metric and response-scale variables with standard deviations in parentheses, proportions shown for dichotomous variables denoted by (d). For comparisons, note that groups (3) and (4) are subgroups of group (2).

F.2 Tests for differences between self-reported degrees of affectedness on environmental variables

Testing for differences	(1) and (2)	(1) and (3)	(1) and (4)	(3) and (4)
	Not affected vs.	Not affected vs.	Not affected vs.	Affected w/o damage
	Affected	Affected w/o damage	Affected w/ damage	vs. Affected w/ damage
ENV variable:				
Impact of Climate Change	0.033	0.069	-0.044	-0.114
	(0.503)	(0.243)	(0.607)	(0.300)
Env. friendly behavior	0.059	0.115*	-0.059	-0.175
	(0.270)	(0.075)	(0.528)	(0.143)
Green Party preference	-0.005	-0.013	0.012	0.026
	(0.855)	(0.691)	(0.797)	(0.656)
Increase in carbon pricing	0.042	0.093	-0.055	-0.148
	(0.674)	(0.451)	(0.748)	(0.479)
No new vehicles with combustion	0.012	0.045	-0.057	-0.101
engines after 2035	(0.898)	(0.699)	(0.735)	(0.600)
Observations	6,043/218	6,043 / 148	6,043 / 70	148 / 70

Differences in means or proportions and p-values for test results in parentheses shown. Two-sided t-tests for metric and response-scale variables, Chi2-tests for dichotomous variables. p-values in parentheses. Significance levels: * p < 0.10, ** p < 0.05, *** p < 0.01

F.3 Tests for differences between self-reported degrees of affectedness on prosociality

Testing for differences	(1) and (2) Not affected vs.	(1) and (3) Not affected vs.	(1) and (4) Not affected vs.	(3) and (4) Affected:	(5) and (6) Damaged: No aid
resultg for unreferences	Affected	affected w/o	affected w/	W/o damage vs.	received vs. aid
		damage	damage	W/ damage	received
Prosociality variable:					
Positive reciprocity	-0.359***	-0.334**	-0.410**	-0.076	0.942*
	(0.003)	(0.020)	(0.047)	(0.789)	(0.071)
Negative reciprocity	0.383**	0.601	0.190	0.792***	-0.446
	(0.024)	(0.113)	(0.351)	(0.007)	(0.514)
Trust	-0.096	-0.019	-0.257	-0.238	-0.590
	(0.556)	(0.924)	(0.365)	(0.478)	(0.310)
Altruism	0.363**	0.531**	0.007	-0.524	0.900
	(0.036)	(0.011)	(0.981)	(0.153)	(0.136)
Dictator game	0.068	0.117	-0.034	-0.151	-0.100
	(0.556)	(0.404)	(0.864)	(0.544)	(0.780)
Past donations (d)	0.003	0.009	-0.011	-0.020	0.109
	(0.918)	(0.784)	(0.827)	(0.734)	(0.271)
Past donations (amount)	72.277*	123.792**	-40.357	-164.149*	-91.059
	(0.072)	(0.010)	(0.567)	(0.069)	(0.471)
Volunteer work (d)	0.069	0.080*	0.047	-0.033	0.186
	(0.044)	(0.054)	(0.438)	(0.654)	(0.127)
Observations	6.043/218	6 043 / 148	6.043 / 70	148 / 70	30/40

Differences in means or proportions and p-values for test results in parentheses shown. Two-sided t-tests for metric and response-scale variables, Chi2-tests for dichotomous variables. (d) denotes dichotomous variables. p-values in parentheses. Significance levels: * p < 0.10, ** p < 0.05, *** p < 0.01

Q. Survey questions

Below, relevant survey questions on the dependent variables (environmental attitudes, environmental behaviors, support for climate policies) are shown. For data set 1, we also show questions on prosociality and the flood experience. Additional questions, such as for measuring individual socioeconomics characteristics as well as all questions in German language, are available upon request from the corresponding author.

Q.1 Survey questions from the interviews that were used to collect data set 1 (RWI Climate-Mobility Panel)

Q.1.1 Environmental attitudes

Impact of Climate Change (V2): What do you think the consequences of climate change will be for your personal living conditions in the coming decades? [Single choice]

- () Very negative consequences
- () Rather negative consequences
- () Roughly equal positive and negative consequences
- () Rather positive consequences
- () Very positive consequences
- () Do not know

Q.1.2 Environmental behavior

Environmentally friendly behavior (V3): Please indicate to what degree you agree with the following statement. [1 - Strongly disagree, 2 - Disagree, 3 - Neither agree nor disagree, 4 - Agree, 5 - Strongly agree, Do not know / no response]

(1) I behave in an environmentally conscious manner, even if this is associated with higher costs and/or efforts.

Green Party preference (S15): Many people lean towards a particular political party for a long time, although they also vote for another party from time to time. What is it like for you? Do you - generally speaking - lean towards a particular party? [Single choice]

() Yes

() No, I am not interested in politics

- () No, because there is no party that appeals to me
- () Do not know / no response

Green Party preference (S16): [Only asked if S15 = Yes] And which party is it? [Single choice]

- () CDU / CSU
- () SPD
- () FDP
- () Bündnis 90 / The Greens
- () The Left Party
- () AfD
- () Pirate Party
- () NPD
- () Another party
- () No response

Q.1.3 Support for climate policies

Increase in carbon pricing (G1): What is your stance on the following measure? [1 - Strongly disagree, 2 - Tend to disagree, 3 - Neither agree nor disagree, 4 - Tend to agree, 5 - Strongly agree, Do not know / no response]

(1) Increase in carbon pricing

No new vehicles with combustion engines after 2035 (C.3): Various measures are currently being discussed in Germany. What is your stance on the following measure? [Random order of responses; 1 - Strongly disagree, 2 - Tend to disagree, 3 - Neither agree nor disagree, 4 - Tend to agree, 5 - Strongly agree, Do not know / no response]

(1) Ban on new registrations of vehicles with combustion engines after 2035

Q.1.4 Prosociality

Positive reciprocity / Negative reciprocity / Trust / Altruism: [Random order of responses] (The Quarterly Journal of Economics, Volume 133, Issue 4, November 2018, Pages 1645–1692, https://doi.org/10.1093/qje/qjy013) How well does each of the following statements describe you as a person? Please use a scale from 0 to 10, with 0 meaning "does not describe me at all" and 10 "describes me perfectly". [Do not know / no response also possible].

- (1) When someone does me a favor, I am willing to return it.
- (2) If I am treated very unjustly, I will take revenge at the first occasion, even if there is a cost to do so.
- (3) I assume that people have only the best intentions.
- (4) I am very willing to give to good causes without expecting anything in return.

Dictator Game donation (G.4 Fairness Experiment):

(Game loosely based on Andreoni and Vesterlund 2001):

The following section is about a game where you and other participants in this survey have the chance to win a Wunschgutschein (Info button: A Wunschgutschein can be redeemed at over 500 stores, including supermarkets, (online) clothing stores, (online) electronics retailers, and much more. More information can be found here: <u>https://www.wunschgutschein.de/</u>).

The voucher amount that you and other participants can receive depends on your decisions and those of the other participants.

You will make a total of six decisions. These decisions do not build on each other and are independent, so you do not need to think about what will come next when making your initial choices.

Specifically, we will provide you with virtual "tokens", each of which corresponds to a certain voucher amount. You will be randomly paired with another study participant, whom we will refer to as your "partner". You will be asked to divide the tokens between yourself and your partner. You and your partner will remain completely anonymous.

For five randomly selected participants, one of the following decisions will be randomly chosen and actually implemented. This means that you will receive a Wunschgutschein worth the tokens you kept for yourself, while your randomly assigned partner will receive a Wunschgutschein worth the tokens you allocated to them.

These vouchers will be sent via email by forsa after the survey is completed.

Game 1: Me – Other Decision 1: Prog: DO NOT RANDOMIZE You have 10 tokens that you can divide between yourself and your assigned partner. Each token is worth 10 euros for both you and your partner. You can choose any distribution (in whole tokens).



Please indicate how many tokens you would like to give to your assigned partner: Decision: ______ tokens (Limit: 0 to 10; no decimals allowed) – Don't know / No response

Past donations (PS1): We would now like to ask about donations. By donations, we mean giving money for social, religious, cultural, charitable, and nonprofit purposes without receiving direct compensation in return. This can include larger amounts as well as smaller ones, such as money put in a donation box. Church collections are also included. Did you donate money last year, that is, in 2021? [Single choice]

- () Yes () No
- () Don't know
- () No response

Past donations (amount) (PS2): [If PS1 = Yes] What was the total amount you donated last year? If you are unsure, please estimate.

- () _____ Euros [Open answer from 0 to 999,999,999]
- () Don't know
- () No response

Volunteer work (PS9): Do you regularly engage in voluntary work (e.g., in a club, foundation, religious community, initiative, etc.)? [Single choice]

- () Yes
- () No
- () Don't know / No response

Q.1.5 Flood experience

Self-reported exposure (F1): *Was your place of residence affected by the flood disaster in July of 2021?* [Single choice]

- () Yes
- () No
- () Do not know / no response

Damage to persons or property in household (F2): [Only asked if F1 = Yes] *Were you or your household directly affected by the flood?* [Multiple choice, except for No]

- [] Yes, with personal injury (physical)
- [] Yes, with personal injury (emotional/psychological)
- [] Yes, with considerable material damage (e.g., to property, house, car, ...)
- [] Yes, with minor material damage (e.g., to property, house, car, ...)
- () No, none of the above
- () Do not know / no response

Aid received during or after flood (F3): [Only asked if F2 = Yes (personal injury and/or material damage)] *Did you receive aid, e.g. with cleaning up, renovations, food, accommodation and/or financial assistance*?

[Multiple choice, random order of responses; only No and Do not know / no response always as last options]

- [] Yes, from public authorities (fire department, German Red Cross, police, city, state, etc.)
- [] Yes, from private entities (private companies, neighbors, friends and relatives, etc.)
- [] Yes, from an insurance company
- [] Yes, via other sources (donations, church, etc.)
- () No, I did not need or accept any aid, although it was offered to me
- () No, I was not offered any aid
- () Do not know / no response

Q.2 Survey questions from the interviews that were used to collect data set 2 (Socio-Ecological Panel)

Q.2.1 Environmental attitudes

Existence of climate change (Klima1): *Based on everything you know; do you think the world's climate is changing or not?* [Single choice]

- () Yes, I believe that the global climate is changing.
- () No, I do not think that the global climate is changing.
- () Do not know / no response

Climate change manmade (Klima2): [Only asked if Klima1 = Yes] Do you think climate change is mostly caused by human activity (as opposed to mostly caused by natural factors)? [Single choice]

- () Mostly by humans
- () Mostly by natural factors
- () Both to the same extent
- () Do not know / no response

Dangers of climate change (ExpB6): [Only asked if Klima1 = Yes] *Do you think climate change is dangerous for humanity?* [Single choice]

- () No, certainly not
- () I am unsure, but rather no
- () I am unsure, but rather yes
- () Yes, certainly
- ($\)$ Do not know / no response

Importance of climate policy (ExpB7): *Do you consider climate policy to be an important field in politics?* [Single choice]

- () No, it is not important
- () I am unsure, but rather no
- () I am unsure, but rather yes
- () Yes, it is important
- () Do not know / no response

Relative importance (Klima3): *In your opinion, what are the two most important problems currently facing Germany?* [Multiple choice, random order of responses; up to two responses possible]

- [] Economic situation
- [] Immigration
- [] Health
- [] Environment and climate change
- [] Unemployment
- [] Rising prices/inflation/cost of living
- [] Terrorism
- [] Crime
- [] Pensions
- [] Taxes
- [] Energy supply
- [] International security situation
- [] Other [Always as last option]
- () Do not know / no response

Q.2.2 Environmental behavior

Green Party preference (SO7): In Germany, many people tend to vote for a particular political party for a long time, although they also vote for another party from time to time. What is it like for you? Do you - generally speaking - lean towards a particular party? And if so, which one? [Single choice]

- () CDU / CSU
- () SPD
- () AfD
- () FDP
- () The Left Party
- () Bündnis 90 / The Greens
- () Another party
- () No party
- () Do not know / no response

Q.2.3 Support for climate policies

No new vehicles with combustion engines after 2030 / Constant increases in tax on petrol and diesel / Speed limit of 130km/h on highways (U12_1): For the mobility transition, there are some measures being discussed that aim to reduce car use and thereby emission of pollutants and carbon dioxide. Do you support the following measures?

[Random order of responses; 1 - No, 2 - More likely no, 3 - Neither, 4 - More likely yes, 5 - Yes, Do not know / no response]

- (1) No new vehicles with combustion engines after 2030
- (2) Constant increase in tax on petrol and diesel fuels, e.g., by 5 Cent per liter
- (3) Speed limit of 130km/h on highways

Q.3 Survey questions from the interviews that were used to collect data set 3 (NEWCOMERS)

Q.3.1 Environmental attitudes

Seriousness of climate change (AB2): (European Commission – Eurobarometer 2020) *How serious a problem do you think climate change is at this moment? Please use a scale from 1 to 10, with 1 meaning it is "not at all a serious problem" and 10 meaning it is "an extremely serious problem".*

(1) Seriousness of climate change

Importance of protecting the environment (AB1): (European Commission – Eurobarometer 2020) *How important is protecting the environment to you personally?*

- () Very important
- () Fairly important
- () Not very important
- () Not at all important

Q.3.2 Environmental behavior

Conserving energy at home (AB4): *How much do you agree or disagree with the following statements?* [1 - *Strongly disagree, 2 - Disagree, 3 - Neither agree nor disagree, 4 - Agree, 5 - Strongly agree*]

(1) When home, I take actions to conserve energy.

Donation to atmosfair (DON. Experiment 3: Donation Experiment)

In the following, we give you the possibility to win $100 \in$. In addition, you have the option to donate part of this potential win.

Many people in developing countries have no access to electricity because their countries cannot afford to expand the expensive power grid. As a result, electricity is often supplied only to densely populated areas, while rural regions have no access.

Instead of electricity, people use for example firewood from the rainforests for cooking. The resulting decline in the forest has a negative impact on the global climate.

So-called Micro-Grids are intended to change this: Instead of waiting for rural regions to be connected to the central power grid, communities can use decentralized power grids. These consist, for example, of a solar panel for electricity generation, a battery for electricity storage and a system of transmission cables that gives all the inhabitants of a village access to the electricity generated. By establishing its own energy supply, the village thus becomes an autonomous "energy community".



Since financial resources are needed to build such Micro-Grids, organisations like the non-profit Atmosfair collect donations to support such projects (here you can find information about an example project in Madagascar: <u>https://www.atmosfair.de/en/climate-protection-projects/solar-energy/madagascar-solar-powered-rural-electrification-program</u>).

On the next screen you have a chance to support such projects by donating to Atmosfair., Every 100^{th} respondent will win 100ϵ , paid out in form of X mingle points. You can decide how much of this possible win should be donated to Atmosfair.

If you win, we will donate your chosen amount to Atmosfair and transfer mingle points worth the remaining amount to you.

Control (Shown randomly to 1/4 of the respondents): *Before you decide, please reflect on the scenario you read above and consider the role of Micro-Grids for electrification in developing countries.*

TREATMENT I (Shown randomly to 1/4 of the respondents): *Before you decide, please imagine a village without power supply. How will the availability of power supply through such a Micro-Grid change the lives of the people living there?*

TREATMENT II (Shown randomly to 1/4 of respondents): *Before you decide, please imagine living in a village without power supply. How will the availability of power supply through such a Micro-Grid change the lives of the people living there?*

TREATMENT III (Shown randomly to 1/4 of the respondents): *Before you decide, please imagine living in a village without power supply. How will the availability of power supply through such a Micro-Grid change your own life?*

Please, spend a few moments reflecting on this question. When you are ready to proceed, click "next". (measure time spent on this page)

How much of $100 \notin$ would you like to donate to Atmosfair? (Info: For every 100^{th} person, we will donate the selected amount to Atmosfair— an NGO that carries out projects to electrify villages. If you are selected, the rest of the amount will be paid to you in form of mingle points. We guarantee you that your decision does not influence your chances to win.)

[_____] \in [allow all numbers from 0 to 100]

The remaining amount is then [100 - donation] \in , which will be paid out to you in form of [Insert the mingle points equivalent of [100 - donation] \in here] mingle points if you have been among the selected respondents.

Note on treatment conditions for study analyzing the 2021 floods: Since all treatment conditions are roughly equally distributed across the sample as well as across the exposed and nonexposed subsamples (that is, about 25% each), we do not further consider in our study the treatment conditions implemented in the survey as they were part of another research.

Q.3.3 Support for climate policies

Coal phase-out / Expansion of electric vehicles / Expansion of public transport (AB6): *The energy transition might comprise a wide range of different policies. Please indicate what your opinion is on the following policy measures.*

[1 - Strongly disagree, 2 - Disagree, 3 - Neither agree nor disagree, 4 - Agree, 5 - Strongly agree]

- Coal phase-out
- Expansion of electric vehicles
- Expansion of public transport

I. Information on availability of analysis code and data

Analysis codes are available from the corresponding author. The data are not yet publicly available but will be made accessible by the FDZ Ruhr in the future.